$C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf$

Chksum: 172110 Cycle 2

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(D) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): January 23, 2004

PURCHASESOFT, INC.

(Exact Name of Registrant as Specified in Charter)

DELAWARE

(State or Other Jurisdiction of Incorporation)

0-11791

13-2897997

(Commission File Number)

(IRS Employer Identification No.)

7514 Girard Ave Ste 1440 La Jolla, CA 92037 (Address of Principal Executive Offices) (Zip Code)

Registrant's telephone number, including area code: **(858) 456-6608**

Merrill Corporation -04-2032-1 Mon Feb 09 12:33:49 2004 Chksum: 91639 Cycle 2 8-K Purchasesoft, Inc.
C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf EDGAR Doc 1 Page 2

In this current report references to "PurchaseSoft" "we," "us," and "our" refer to PurchaseSoft, Inc.

FORWARD LOOKING STATEMENTS

This current report contains certain forward-looking statements and any statements contained in this current report that are not statements of historical fact may be deemed to be forward-looking statements. Without limiting the foregoing, words such as "may," "will," "expect," "believe," "anticipate," "estimate" or "continue" or comparable terminology are intended to identify forward-looking statements. These statements by their nature involve substantial risks and uncertainties, and actual results may differ materially depending on a variety of factors, many of which are not within PurchaseSoft's control. These factors include, but are not limited to, economic conditions generally and in the markets in which PurchaseSoft may participate, competition within PurchaseSoft's markets and failure by PurchaseSoft to successfully develop business relationships.

ITEM 2: ACQUISITION AND DISPOSITION OF ASSETS

Acquisition of Computer Information Enterprises, Inc.

General

On January 23, 2004 PurchaseSoft entered into an Asset Purchase Agreement ("Purchase Agreement") with Computer Information Enterprises, Inc. ("CIE"), a California corporation. Pursuant to the Purchase Agreement, PurchaseSoft agreed to purchase, subject to the terms and conditions of the Purchase Agreement, certain assets used by CIE in its software business (the "Software Business"). In addition to the purchase of assets from CIE, PurchaseSoft will, at the closing of the acquisition, enter into a "License Agreement" with CIE whereby PurchaseSoft will have the exclusive, worldwide license to commercialize certain intellectual property owned by CIE, which is used in the Software Business. The closing of the Purchase Agreement is subject to several conditions, including but not limited to, approval by CIE's stockholders and accordingly, there can be no assurance that the agreed upon acquisition will be completed. If all conditions of the Purchase Agreement are fulfilled or waived, we anticipate that the Purchase Agreement will close in February 2004.

Description of Business

We have agreed to purchase assets and licensing technologies from CIE that will, upon closing, allow us to operate the Software Business. We intend, through our operations of the Software Business, to develop and market products that provide businesses with a document imaging solutions that is integrated with a customer's Solomon or Great Plains ERP software.

CIE's primary products, ImageLink and AP Distar are fully developed and both were introduced for sale by CIE to the market within the past two years. The products are sold through a reseller network, primarily in the US. The Company operates from leased facilities in Irvine, California and has seven employees. It will operate as a division of PurchaseSoft.

General Terms of the Purchase Agreement

The Purchase Agreement provides that PurchaseSoft will acquire from CIE, the assets relating to the Software Business for the following considerations: (i) the issuance of approximately 1,000,000 shares of our common stock to CIE; and (ii) the assumption of certain liabilities relating to the Software

Merrill Corporation -04-2032-1 Mon Feb 09 12:33:49 2004 Chksum: 1043311 Cycle 2 8-K Purchasesoft, Inc.

EDGAR

Doc 1 Page 3

Business and the assets purchased. The closing of the purchase is subject to several conditions including the approval of the transaction by the stockholders of CIE.

The acquisition agreement contains customary representations and warranties relating to each company's corporate status, corporate authority to complete the acquisition, capital structure and corporate conduct prior to the closing. Each company provided corporate documentation to the other for due diligence purposes. Also, CIE agreed to conduct its business in the normal course, and not to sell, pledge, or assign any assets, amend its articles of incorporation or bylaws prior to the closing of the transaction.

Currently 7 employees of CIE are employed in the Software Business. One of these employees will retain employment with CIE and we will offer the remaining employees employment in our operation of the Software Business.

A copy of the Purchase Agreement is attached hereto as an exhibit.

License Agreement

C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf

In addition to the assets we are purchasing from CIE, we are also licensing certain "Intellectual Property" from CIE. The Intellectual Property we are licensing includes any and all patents, patent registrations, patent applications, data rights, utility models, business processes, trademarks, trade secrets, know how, trade names, registered or unregistered designs, mask works, copyrights, moral rights and any other form of proprietary protection afforded by law to intellectual property, or any applications therefore, which arises or is enforceable under the laws of the United States, any other jurisdiction or any bi-lateral or multi-lateral treaty regime which relate to the Software Business.

The License Agreement, which will be effective on the date of closing of Purchase Agreement, requires us to pay a royalty of 10% of gross revenues (as defined in the License Agreement) for product sales transactions. The License Agreement grants us the worldwide, exclusive license to use the Intellectual Property to manufacture, market, sublicense or otherwise commercialize products using the technology included in the Intellectual Property. The License Agreement provides for minimum annual royalties \$30,000 per year.

The royalties agreed to in the License Agreement are to be paid until the earlier of the date on which we have paid CIE an aggregate of One Million Dollars (\$1,000,000) or the fifth anniversary of effective date of the License Agreement. ("Total Royalty"). If we pay CIE the Total Royalty, CIE is required to assign and transfer to us all of CIE's rights, title and interests in and to the Intellectual Property for no additional consideration.

If we fail to pay the required minimum royalties are otherwise breach the License Agreement, CIE may terminate the license.

Consideration for the Acquisition

The consideration we are to pay CIE pursuant to the Purchase Agreement was negotiated at "arms length" and our management relied on representations made by CIE's management and other documents and information provided to us. Our management considered factors used in similar proposals to determine the amount of consideration appropriate for the acquisition of the assets. These factors included the relative value of the assets, the present and past operations of the Software Business, the future potential of the Software Business, the continued employment of employees of the Software Business, and the potential benefit of the transaction to the stockholders of PurchaseSoft.

Merrill Corporation -04-2032-1 Mon Feb 09 12:33:49 2004 Chksum: 667228 Cycle 2 8-K Purchasesoft, Inc.
C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf EDGAR Doc 1 Page 4

Our board of directors determined that the terms of the Purchase Agreement and License Agreement are reasonable based upon the above factors. Our board did not seek a third party fairness opinion or any valuation or appraisal of the terms of the transaction. Thus, our stockholders will not have the benefit of a third party opinion that the terms of the Purchase Agreement and License Agreement are fair from a financial point of view.

Interests of Certain Persons

Other than as described in this report, there have been no contacts, negotiations or transactions within the past two years between PurchaseSoft or any of our directors, executive officers or their affiliates, on the one hand, and CIE or its affiliates, on the other hand, regarding the acquisition, consolidation, acquisition of shares or election of directors.

Acquisition of Netshare Solutions GmbH

General

On January 30, 2004 PurchaseSoft entered into a Purchase Agreement ("Purchase Agreement") with Netshare Solutions GmbH and the shareholders of Netshare Solutions GmbH ("Netshare"), a German company. Pursuant to the Purchase Agreement, PurchaseSoft agreed to purchase, subject to the terms and conditions of the Purchase Agreement, all of the stock of Netshare, effective on that date.

Description of Business

We have agreed to purchase all of the stock of Netshare that will, upon closing, allow us to operate the company as a wholly owned subsidiary of PurchaseSoft. We intend, through our operations of the business, to develop and market products that provide businesses with an eCommerce platform for procurement, sourcing and ebusiness based on Microsoft's .NET software.

Netshare's primary product, FlexTrade is fully developed and was introduced for sale by Netshare to the market within the past two years. The products are sold direct, primarily in Germany. The Company operates from leased facilities in Hanover Germany and has seventeen employees. The predecessor company, Netshare AG was forced into insolvency in November 2003. Netshare Solutions GmbH was formed in December 2003 by the management team of Netshare AG to acquire some of the assets in order to continue service and support to customers.

General Terms of the Purchase Agreement

The Purchase Agreement provides that PurchaseSoft acquire from Netshare's shareholders, 100% of the stock of the company for the following considerations: (i) the issuance of 8,000,000 shares of our common stock to Netshare's shareholders; (ii) a cash payment of €13,000 to one outside shareholder; and (iii) payment of £120,000 to the managing shareholders over a period of approximately one year contingent on the achievement of certain revenue milestones.

The acquisition agreement contains customary representations and warranties relating to each company's corporate status, corporate authority to complete the acquisition, capital structure and corporate conduct prior to the closing. Each company provided corporate documentation to the other for due diligence purposes. Also, Netshare had agreed to conduct its business in the normal course, and not

Merrill Corporation -04-2032-1 Mon Feb 09 12:33:49 2004 Chksum: 117415 Cycle 2 8-K Purchasesoft, Inc.
C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf EDGAR Doc 1 Page 5

to sell, pledge, or assign any assets, amend its articles of incorporation or bylaws prior to the closing of the transaction.

Currently Netshare has 17 employees. We will retain all of these employees, including the four founders of Netshare, and will maintain the offices in Hanover Germany. The subsidiary will be managed by Werner Loechle, COO of PurchaseSoft and based in Stuttgart Germany.

A copy of the Purchase Agreement is attached hereto as an exhibit.

Consideration for the Acquisition

The consideration we paid Netshare pursuant to the Purchase Agreement was negotiated at "arms length" and our management relied on representations made by Netshare's management and other documents and information provided to us. Our management considered factors used in similar proposals to determine the amount of consideration appropriate for the acquisition of the company. These factors included the relative value of the assets, the present and past operations of the company, the future potential of the company, the continued employment of employees, and the potential benefit of the transaction to the stockholders of PurchaseSoft.

Our board of directors determined that the terms of the Purchase Agreement are reasonable based upon the above factors. Our board did not seek a third party fairness opinion or any valuation or appraisal of the terms of the transaction. Thus, our stockholders will not have the benefit of a third party opinion that the terms of the Purchase Agreement are fair from a financial point of view.

Interests of Certain Persons

Other than as described in this report, there have been no contacts, negotiations or transactions within the past two years between PurchaseSoft or any of our directors, executive officers or their affiliates, on the one hand, and Netshare or its affiliates, on the other hand, regarding the acquisition, consolidation, acquisition of shares or election of directors.

ITEM 5: OTHER EVENTS AND REGULATION FD DISCLOSURE

Private Placement & Bridge Loan Financing

In order to complete the acquisitions described above Company will raise capital in the form of bridge loans and a private placement of common stock. The Company will also be required to issue warrants for its common shares in order to compensate individuals for the risk associated with extending credit to the Company.

Management received the authorization of the Board to borrow \$325,000 in the form of bridge loans with a term of up to six months, at an interest rate of 6% per annum with the Company issuing warrants for up to 2,895,000 shares at a price per share of \$.01to \$.0001 each. The use of proceeds is for the acquisitions, acquisition expenses and working capital.

Management also received authorization for a private placement of Company stock to raise up to \$800,000. The Company has identified several accredited investors, including the Company's President, Steve Flagg and the Company's VP International Werner Loechle. The Offering was required for and contingent on the completion of the CIE and Netshare acquisitions. The offering is for up to 40,000,000 shares at \$0.02 per share. The offering will terminate no later than March 31, 2004.

 Merrill Corporation -04-2032-1
 Mon Feb 09 12:33:49 2004
 Chksum: 860553 Cycle 2

 8-K
 Purchasesoft, Inc.

 C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf
 EDGAR
 Doc 1 Page 6

Increase in Authorized Shares of Common Stock

The board has voted to authorize the increase of common shares outstanding from 50,000,000 to 100,000,000. The amendment to the articles of incorporation has been filed with the State of Delaware.

Appointment of Officers, Directors and Management Compensation

With the completion of the clean up and restart of Company and the execution of agreements for the CIE and Netshare acquisitions, Tom Marsh will resign his positions as Secretary, Treasurer and acting CFO, effective February 10th, to make room for the Company to retain and pay new officers required to operate the new businesses. Marsh will continue to serve as a Director of the Company until August 2004. As a part of the transition agreement, Marsh has agreed to sell 5,271,210 shares of his PS stock to Flagg at a price per share of \$.0001.

The Board has appointed Kevin Yanoscik to serve as the Company's CFO and member of the Board of Directors. His responsibilities will be to focus on the financial management of the Company and the operation of the new CIE division. He will receive \$3,000 per month and will receive a warrant for 1,000,000 shares of Company stock at the price of \$.0001 per share.

Kevin A. Yanoscik – Mr. Yanoscik comes to the Company after working at Marsh+Flagg since 2002, as a financing and merger and acquisition advisor. Kevin Yanoscik joined the firm after a twenty-year career in banking. Starting with the predecessor banks to the current Union Bank of California, Kevin started his career in Seattle, Washington before relocating to San Diego and then Los Angeles, CA. He has also worked for Hong Kong Shanghai Bank and spent several years with NationsBank, now Bank of America, and LaSalle Business Credit, part of the ABN AMRO Group. With a diverse career including positions in credit and marketing, Kevin has worked in middle-market commercial banking and also asset-based lending and structured finance/cash flow lending. He brings extensive transactional experience in mergers and acquisitions, turnarounds, and has had extensive training in trade finance. Kevin received a BA degree from Temple University in Philadelphia, PA in political science and an MBA/Master of International Management Degree from the University of Denver.

The Board has appointed Werner Loechle to serve as the Company's COO and member of the Board of Directors. He will also be a member of the Supervisory Board of Netshare Solutions GmbH. His responsibilities will be to focus on the management of the Company's German subsidiary and all International operations. He will be compensated through Netshare at the rate of &10,000 per month plus bonuses and will receive a warrant for 4,000,000 shares of Company stock at the price of &0001 per share.

Werner Kuno Loechle – Mr. Loechle joined PurchaseSoft in August 2002, bringing considerable high tech and pharmaceutical experience and a broad network of clients and contacts in the European business community. Werner is responsible for the firm's European client development and sourcing of prospective acquisitions. Werner is now or has been a member of the Board of a New Economy Internet Company, Chief Executive Officer of a New Economy Internet company, founder and President of Profund Management Consulting Ltd, Senior Vice President, KnowledgeWare Worldwide Inc, Paris, President and Managing Partner in Ernst &

Merrill Corporation -04-2032-1 Mon Feb 09 12:33:49 2004 Chksum: 235807 Cycle 2 8-K Purchasesoft, Inc.
C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf EDGAR Doc 1 Page 7

Young Consulting Ltd Stuttgart, and Ernst & Young CASE Services (pan European software- and service organization) in Paris. He was also CIO in a global Pharmaceutical company, CIO in an international Pipeline Engineering company (Germany and Saudi-Arabia), Scientist at the European Center for Medium Range Weather Forecasts (ECMWF), Shinfield Park/UK, and Scientific Associate at the pan-European nuclear physics research laboratory (CERN), Geneva/CH. Werner has published numerous articles and is a public speaker.

As a part of the changes required by the acquisitions, Flagg has been appointed CEO, serving the Company full-time and will be compensated at the rate of \$10,000 per month. He will receive a warrant for 4,000,000 shares of Company stock at the price of \$.0001 per share.

During the past seven months, Steven Flagg, CEO and Thomas Marsh, CFO have expended substantial time and effort on behalf of the Company in pursuit of acquisitions and in the operation of the business. As such, and in the best interest of the shareholders to continue this effort, the Company has advanced funds to Flagg and Marsh, against future royalty and other compensation to be earned under their contracts with the Company. Those payments have been expensed by PS as consulting expenses during the periods such payments were made as reported in the Company's 10-QSB filings. The Board hereby authorizes such payments and Flagg and Marsh each individually accept those payments, as paid or accrued, as full and complete payment for their services through the period ending January 31, 2004.

ITEM 7: FINANCIAL STATEMENTS, PRO FORMA FINANCIAL INFORMATION AND EXHIBITS

(a) Financial Statements: N/A

(b) Proforma Financial Statements: N/A

(c) Exhibits.

Exhibit No.	Title
2.1	Asset Purchase Agreement CIE
2.2	Purchase Agreement Netshare Solutions GmbH
99.1	Press Release

ITEM 9. REGULATION FD DISCLOSURE

Attached here to as Exhibit 99.1 is a Press Release regarding the execution of the Asset Purchase Agreement described in Item 2 of this Form 8-K.

Merrill Corporation -04-2032-1 Mon Feb 09 12:33:49 2004 Chksum: 1020348 Cycle 2 8-K Purchasesoft, Inc.

C:\jms\jpatrow\04-2032-1\task71916\2032-1-ba.pdf

EDGAR

Doc 1 Page 8

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

PURCHASESOFT, INC.

/s/ Thomas B. Marsh

Signature

Name: Thomas B. Marsh Title: Secretary and Treasurer

Dated: February 5, 2004

 Merrill Corporation -04-2032-1
 Sat Feb 07 01:03:32 2004
 Chksum: 61851 Cycle 1

 EX-2.1
 Purchasesoft, Inc.

 C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf
 EDGAR
 Doc 2 Page 1

Exhibit 2.1

ASSET PURCHASE AGREEMENT

by and among

${\bf Computer\ Information\ Enterprises,\ Inc.}$

and

PurchaseSoft, Inc.

dated January 23, 2004

TABLE OF CONTENTS

EDGAR

Doc 2 Page 2

ARTICLE I DEFINITIONS

- **Defined Terms**
- $\frac{1.1}{1.2}$ Construction of Certain Terms and Phrases

ARTICLE II PURCHASE AND SALE OF ASSETS

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

- Purchase and Sale of Certain Assets of the Company
- 2.1 2.2 2.3 2.4 Excluded Assets
- Assumed Liabilities/Excluded Liabilities
- Purchase Price
- Allocation of Aggregate Purchase Price
- 2.5 2.6 2.7 Private Placement
- Sales, Use and Other Taxes
- 2.8 2.9 Bulk Sales Compliance
- Closing

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

- 3.1 3.2 3.3 Organization of the Company
- Authority
- Financial Statements and Reports
- <u>3.4</u> No Conflicts
- 3.5 Consents, Approvals and Filings
- No Undisclosed Liabilities 3.6
- <u>3.7</u> Purchased Assets
- 3.8 Benefit Plans; ERISA
- Real Property
- 3.9 3.10 Intellectual Property Rights.
- 3.11 Litigation
- 3.12 3.13 Compliance with Law
- Contracts.
- <u>3.14</u> **Environmental Matters**
- 3.15 **Inventory**
- 3.16 Plants, Buildings, Structures, Facilities and Equipment
- <u>3.17</u> Customer Lists and Accounts
- 3.18 Relationships with Suppliers and Licensors
- 3.19 Tax Matters
- <u>3.20</u> **Permits**
- **Brokers**
- 3.21 3.22 3.23 Material Misstatements and Omissions
- **Investment Representations**

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PURCHASER

- <u>4.1</u> <u>4.2</u> Organization
- Capital Stock of Purchaser

8.5

Further Assurances

Merrill Corpo	oration -04-2032-1 Sat Feb 07 01:03:32 2004		Chksum: 626425 Cycle 1
EX-2.1	Purchasesoft, Inc.		
C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf		EDGAR	Doc 2 Page 4
	, <u>, , , , , , , , , , , , , , , , , , </u>		
9.6	Denote Heden Consider France Act of 1024		
8.6	Reports Under Securities Exchange Act of 1934		
8.7	Transfer of Rights to Name		
<u>8.8</u>	Access to Records		
<u>8.9</u>	<u>Transitional Use of Financial System</u>		
ARTICLE I	X MISCELLANEOUS		
<u>9.1</u>	Termination.		
9.2	Notices		
9.2 9.3 9.4 9.5 9.6 9.7 9.8	Entire Agreement		
9.4	Waiver		
9.5	Amendment		
9.6	No Third Party Beneficiary		
$\overline{9.7}$	No Assignment; Binding Effect		
${9.8}$	Headings		
9.9	Severability		
9.10	Governing Law		
9.11	Consent to Jurisdiction and Forum Selection		
9.12	Expense		
	•		
9.13	Construction		
<u>9.14</u>	<u>Counterparts</u>		

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 933553 Cycle 1

EX-2.1 Purchasesoft, Inc. $C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf$

Schedules and Exhibits

EDGAR

Doc 2 Page 5

Schedules

Products. Schedule 2.1(a) Schedule 2.1(b) Schedule 2.1(c) Schedule 2.1(d) Schedule 2.1(e) Property

Customer Lists and Accounts

Assumed Contracts

Permits

Schedule 2.2 Schedule 2.3 **Excluded Assets** Assumed Liabilities

Schedule 2.5 Allocation

Disclosure Schedule

Exhibits

Exhibit A License Agreement Bill of Sale

Exhibit B
Exhibit C General Assignment

Registration Rights Agreement Exhibit D

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 635102 Cycle 1 EX-2.1 Purchasesoft, Inc.

ASSET PURCHASE AGREEMENT

EDGAR

Doc 2 Page 6

This Asset Purchase Agreement ("Agreement") is made and entered into as of January 23, 2004, by and among PurchaseSoft, Inc., a Delaware corporation ("Purchaser") and Computer Information Enterprises, Inc., a California corporation (the "Company").

RECITALS

WHEREAS, Purchaser desires to acquire from the Company, and the Company desires to sell to Purchaser, certain assets of the Company's software business (the "Business") on the terms and subject to the conditions set forth in this Agreement; and

WHEREAS, the Purchaser desires to assume certain liabilities of the Business as more fully described herein; and

WHEREAS, the Company shall retain certain product rights with respect to the products and assets of the Business, which product rights will be exclusively licensed to Purchaser pursuant to the terms of the License Agreement; and

WHEREAS, certain other assets and liabilities of the Company will remain with the Company and not be transferred to Purchaser, all as more particularly set forth herein; and

WHEREAS, Purchaser and the Company acknowledge that this Agreement and the agreements attached as exhibits hereto collectively constitute the agreements necessary to accomplish the transactions contemplated by this Agreement and are parts of an integrated arrangement between the parties with respect to the purchase and sale of the Purchased Assets (as defined in Section 1.1) and certain licenses and other relationships between the parties, and that separate agreements have been used for the sake of convenience.

NOW, THEREFORE, in consideration of the premises and the mutual covenants and promises contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE I DEFINITIONS

- 1.1 **Defined Terms.** As used in this Agreement, the following defined terms have the meanings indicated below:
- "Actions or Proceedings" means any action, suit, proceeding, arbitration, Order, inquiry, hearing, assessment with respect to fines or penalties or litigation (whether civil, criminal, administrative, investigative or informal) commenced, brought, conducted or heard by or before, or otherwise involving, any Governmental or Regulatory Authority.
- "Affiliate" means, with respect to any Person, another Person that directly, or indirectly through one or more intermediaries, controls, is controlled by or is under common control with such Person.
- "Assets and Properties" and "Assets or Properties" of any Person each means all assets and properties of every kind, nature, character and description (whether real, personal or mixed, whether tangible or intangible, whether absolute, accrued, contingent, fixed or otherwise and wherever situated), including the goodwill related thereto, operated, owned or leased by such Person, including, without limitation, cash, cash equivalents, accounts and notes receivable, chattel paper, documents, instruments, general intangibles, real estate, equipment, inventory, goods and Intellectual Property.
- "Assumed Contracts" has the meaning set forth in Section 2.1(d) below.

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

"Assumed Liabilities" has the meaning set forth in Section 2.3(a) below.

"Books and Records" of any Person means all files, documents, instruments, papers, books, computer files (including but not limited to files stored on a computer's hard drive or on floppy disks), electronic files and records in any other medium relating to the business, operations or condition of such Person.

Chksum: 27768 Cycle 1

- "Business Day" means a day other than Saturday, Sunday or any day on which banks located in the State of California are authorized or obligated to close.
- "Cash Purchase Price" means the amount of the cash to be paid by Purchaser at Closing.
- "Claim Notice" has the meaning set forth in Section 8.2(c).
- "Closing" has the meaning set forth in Section 2.9(a) below.
- "Closing Date" has the meaning set forth in Section 2.9(a) below.
- "Closing Shares" has the meaning set forth in Section 2.4(a) below.
- "Code" means the Internal Revenue Code of 1986, as amended.
- "Company" has the meaning set forth in the first paragraph of this Agreement.
- "Company Disclosure Schedule" means the disclosure schedule of the Company attached hereto which sets forth the exceptions to the representations and warranties contained in <u>Article III</u> hereof and certain other information called for by this Agreement.
- "Company Intellectual Property" means any Intellectual Property exclusively relating to the conduct of the Business that is owned by, exclusively licensed to or managed by the Company.
- "Company Records" has the meaning set forth in Section 2.1(f) below.
- "Company Reports" has the meaning set forth in Section 3.3 below.
- "Confidentiality Agreement" has the meaning set forth in Section 5.11 below.
- "Customer Lists and Accounts" has the meaning set forth in Section 2.1(f) below.
- "Damages" has the meaning set forth in Section 8.2(a) below.
- "Dispute Notice" has the meaning set forth in Section 8.2(c).
- "Encumbrances" means any mortgage, pledge, assessment, security interest, deed of trust, lease, lien, adverse claim, levy, charge or other encumbrance of any kind, or any conditional sale or title retention agreement or other agreement to give any of the foregoing in the future.
- "Environment" means any surface water, ground water, drinking water supply, land surface or subsurface strata, ambient air and any indoor workplace.
- "Environmental Laws" means all national, state, local and foreign laws, codes, regulations, common law, requirements, directives, Orders, and administrative or judicial interpretations thereof, all as in effect on the date hereof or on the Closing Date, that may be enforced by any Governmental or Regulatory Authority, relating to pollution, the protection of the Environment and the safety of workers and the public, or the regulation of the emission, discharge, disposal, release or threatened release of Materials in or into the Environment.
- "Environmental Notice" means any written notice by any Person alleging potential liability (including, without limitation, potential liability for investigatory costs, cleanup costs, governmental costs, harm or damages to person, property, natural resources or other fines or penalties) arising out of, based on or resulting from (a) the emission, discharge, disposal, release or threatened release in or into the Environment of any Materials or (b) circumstances forming the basis of any violation, or alleged violation, of any applicable Environmental Law.

EX-2.1 Purchasesoft, Inc.

Chksum: 915710 Cycle 1

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.

- "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Excluded Assets" has the meaning set forth in Section 2.2 below.
- "Fair Market Value" has the meaning set forth in Section 8.2(e)(iv) below.
- "Financial System" has the meaning set forth in Section 8.9 below.
- "GAAP" means United States generally accepted accounting principles.
- "General Assignment" has the meaning set forth in Section 2.9(b)(iv) below.
- "Governmental or Regulatory Authority" means any court, tribunal, arbitrator, authority, agency, commission, official or other instrumentality of the United States or other country, any state, county, city or other political subdivision.
- "Indemnity Shares" means 50,000 of the Closing Shares to be held by the Company in accordance with the terms of this Agreement.
- "Intellectual Property" means (i) trademarks, service marks, trade names, trade dress, labels, product configurations, logos, and all other names and slogans associated with any products or embodying the goodwill of the Business, whether or not registered, and any applications or registrations therefor (including any goodwill or common law rights associated therewith), (ii) plans, design drawings, specifications and performance criteria, operating instructions and maintenance manuals, manufacturing information (including production documentation, methods, layouts and supplier and cost information), copies of on-site computer software and related documentation (including, without limitation, source and object code to the extent available), prototypes, models or samples, ideas, concepts and data, research records, all promotional literature, customer and supplier lists and similar data and information and all other confidential or proprietary technical and business information, (iii) copyrights, copyright registrations and applications for registration, (iv) patent and patent applications (including all reissues, divisions, continuations, continuations-in-part, renewals, and extensions of the foregoing) owned by the Company and (v) all other intellectual property rights and goodwill related thereto.
- "Knowledge of the Company" or "Known to the Company" means the actual knowledge of any Company executive officer after reasonable inquiry.
- "Leased Real Property" has the meaning set forth in Section 3.9 below.
- "Liabilities" means any liability, debts, obligations of any kind or nature (whether known or unknown, whether asserted, or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, and whether due or to become due), including but not limited to any liability for Taxes.
- "License Agreement" has the meaning set forth in Section 2.2 below.
- "Materials" means pollutants, contaminants or chemical, industrial, hazardous, radioactive or toxic materials or wastes, including, without limitation, petroleum, petroleum products and radiation.
- "Material Adverse Effect" means, for any Person, a material adverse effect, whether individually or in the aggregate, (a) on the business, operations, financial condition, Assets and Properties, Liabilities or prospects of such Person, or (b) on the ability of such Person to consummate the transactions contemplated hereby. For an event or condition to have a Material Adverse Effect hereunder, such event or condition must specifically and particularly affect the Person in question as opposed to affecting generally the economy, an industry or society (or a segment thereof).
- "Offer Recipients" has the meaning set forth in Section 5.9(a) below.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004

EX-2.1 Purchasesoft Inc

EX-2.1 Purchasesoft, Inc.
C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf EDGAR Doc 2 Page 9

Chksum: 651693 Cycle 1

"Order" means any writ, judgment, decree, injunction or similar order of any Governmental or Regulatory Authority (in each such case whether preliminary or final).

- "Ordinary Course of Business" means the action of a Person that is consistent with the past practices of such Person and is taken in the ordinary course of the normal day-to-day operations of such Person.
- "Permits" means all licenses, permits, certificates of authority, authorizations, approvals, registrations and similar consents granted or issued by any Governmental or Regulatory Authority relating to the Business, the Purchased Assets or the Assumed Liabilities.
- "Permitted Encumbrances" means (i) any Encumbrance for taxes that are not yet due or payable, (ii) any Encumbrance for tax assessments and other charges or claims with respect to taxes, the validity of which are being contested in good faith by appropriate proceedings for which adequate reserves have been established in accordance with generally accepted accounting principles, (iii) any minor imperfection of title or similar Encumbrances which individually or in the aggregate with other such Encumbrances does not impair the value of the property subject to such Encumbrance or the use of such property in the conduct of the Business, (iv) mechanics' and materialmen's liens incurred in the Ordinary Course of Business for construction or alterations, (v) statutory liens of landlords and workmen's, repairmen's, warehousemen's and carriers' liens arising in the Ordinary Course of Business, (vi) requirements incurred or other Encumbrances relating to deposits made in the Ordinary Course of Business in connection with workers' compensation, unemployment insurance and other similar statutory requirements and (vii) Encumbrances constituted by the terms of (A) any equipment lease; (B) any capital lease; (C) any license and (D) any real property lease.
- "**Person**" means any natural person, corporation, general partnership, limited partnership, limited liability company, proprietorship, other business organization, trust, union, association or Governmental or Regulatory Authority.
- "Products" has the meaning set forth in Section 2.1(a) below.
- "**Property**" has the meaning set forth in Section 2.1(b) below.
- "Purchased Assets" has the meaning set forth in Section 2.1 below.
- "Purchaser" has the meaning set forth in the first paragraph of this Agreement.
- "Purchaser Disclosure Schedule" means the disclosure schedule of the Purchaser attached hereto which sets forth the exceptions to the representations and warranties contained in Article IV hereof and certain other information called for by this Agreement.
- "Purchaser Group" has the meaning set forth in Section 8.2(a) below.
- "Registration Rights Agreement" means the agreement substantially in the form of Exhibit D attached hereto.
- "SEC" means the United States Securities and Exchange Commission, or any successor entity.
- "Securities" means, collectively, the Closing Shares and Indemnity Shares.
- "Securities Act" means the Securities Act of 1933, as amended.
- "Survival Period" has the meaning set forth in Section 8.1 below.
- "Tax" (and, with correlative meaning, "Taxes," "Taxable" and "Taxing") means all sales and use taxes, real and personal property taxes, gross receipts taxes, income taxes, documentary transfer taxes, employment taxes, withholding taxes, unemployment insurance contributions and other taxes or governmental charges of any kind, however denominated, including any Liability in respect thereto, under any federal, state, local, foreign or other applicable tax law.

EX-2.1 Purchasesoft, Inc.

"Tax Return" means any return, report, information return, schedule or other document (including any related or supporting information) filed or required to be filed with respect to any taxing authority with respect to Taxes.

- "Updated Capitalization Representation" has the meaning set forth in Section 4.2 below.
- "Updated Company Disclosure Schedule" has the meaning set forth in Section 5.13 below.
- "Updated Purchaser Disclosure Schedule" has the meaning set forth in Section 5.13 below.
- 1.2 Construction of Certain Terms and Phrases. Unless the context of this Agreement otherwise requires, (a) words of any gender include each other gender; (b) words using the singular or plural number also include the plural or singular number, respectively; (c) the terms "hereof," "herein," "hereby" and derivative or similar words refer to this entire Agreement; (d) the terms "Article" or "Section" refer to the specified Article or Section of this Agreement; (e) the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or"; and (f) "including" means "including without limitation." Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP.

ARTICLE II PURCHASE AND SALE OF ASSETS

- 2.1 **Purchase and Sale of Certain Assets of the Company.** Subject to the terms and conditions hereof, the Company shall sell, assign, grant, transfer, convey and deliver to Purchaser, and Purchaser shall purchase and accept from the Company as of the Closing, other than the Excluded Assets, all of the assets of the Business, wherever situated, as the same shall exist on the Closing Date (collectively, the "Purchased Assets"), including without limitation the following:
- (a) <u>Products</u>. All products listed on <u>Schedule 2.1(a)</u>, including without limitation all specifications, documentation, supplements, improvements, modifications, updates, corrections and enhancements to past versions of such products, shipping versions of such products and versions of such products currently under development (the "Products")
- (b) <u>Property</u>. All fixed assets and tangible personal property used exclusively in the operation of the Business and set forth in <u>Schedule 2.1(b)</u> attached hereto (the "Property");
- (c) <u>Customer Lists</u>. All customer lists and customer accounts owned by the Company and used primarily in operation of the Business and set forth in <u>Schedule 2.1(c)</u> attached hereto (the "Customer Lists and Accounts");
- (d) <u>Assumed Contracts</u>. All rights of the Company under the agreements entered into between the Company and third parties named therein in the operation of the Business and listed in <u>Schedule 2.1(d)</u> attached hereto (the "Assumed Contracts");
- (e) <u>Permits</u>. All Permits (other than Permits that are not assignable pursuant to applicable laws) issued to or held by the Company relating to the Purchased Assets, the Business or Assumed Liabilities as forth in <u>Schedule 2.1(e)</u> attached hereto;
- (f) <u>Company Records</u>. All of the Books and Records of the Company exclusively related to the Business, Purchased Assets and Assumed Liabilities (the "Company Records");
- 2.2 **Excluded Assets.** The Company shall retain all of its rights, title and interest in and to all the assets of the Company other than the Purchased Assets including, without limitation, the Company Intellectual Property and the assets set forth in Schedule 2.2 (collectively, the "Excluded Assets"). Certain Company Intellectual Property shall be licensed to Purchaser in exchange for royalty payments

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 546803 Cycle 1 EX-2.1 Purchasesoft, Inc.

pursuant to the terms of a License Agreement to be executed in connection herewith, substantially in the form of $\underline{\text{Exhibit A}}$ hereto (the "License Agreement").

EDGAR

Doc 2 Page 11

2.3 Assumed Liabilities/Excluded Liabilities.

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

- (a) As of the Closing Date, Purchaser agrees to assume, satisfy or perform when due those liabilities and obligations of the Company listed in <u>Schedule 2.3</u> (the "Assumed Liabilities").
- (b) Other than the Assumed Liabilities, Purchaser shall not assume, or be deemed to have assumed or guaranteed, or otherwise be responsible for any liability, obligation or claims of any nature of the Company, whether matured or unmatured, liquidated or unliquidated, fixed or contingent, known or unknown, or whether arising out of acts or occurrences prior to, at or after the date hereof. Without limiting the generality of the foregoing, the Company shall remain liable for all, Tax liabilities, litigation matters involving the Company and the payment of all Liabilities and obligations to personnel of the Company with respect to the notice and continuation coverage requirement of Section 4980B(e) of the Code and regulations thereunder, payroll, overtime, accrued vacation time, holiday time, severance arrangements or worker's compensation of any nature which are accrued but unpaid as of the Closing Date.
 - 2.4 **Purchase Price.** On the Closing Date, as consideration for the Purchased Assets, Purchaser agrees:
- (a) To deliver to the Company a certificate evidencing 1,000,000 shares of common stock of Purchaser issued in the name of the Company (the "Closing Shares");
 - (b) To pay the royalties described in the License Agreement;
 - (c) To assume the Assumed Liabilities.
- 2.5 **Allocation of Aggregate Purchase Price.** The allocation of the purchase price set forth in Section 2.4 above shall be as set forth on <u>Schedule 2.5</u> attached hereto. Purchaser and the Company agree (a) to report the sale of the Purchased Assets for federal and state Tax purposes in accordance with the allocations set forth on <u>Schedule 2.5</u> hereto and (b) not to take any position inconsistent with such allocations on any of their respective tax returns.
- 2.6 **Private Placement.** The Closing Shares to be issued to the Company will be exempt from the registration requirements of the Securities Act pursuant to the private placement exemption provided by Rules 505 and/or 506 of Regulation D promulgated under the Securities Act and/or Section 4(2) of the Securities Act, pursuant to rule 144, and applicable state securities laws, based in part upon the representations and warranties of the Company contained herein. The Company hereby agrees to take all actions and execute all subscription and other documents which Purchaser reasonably deems necessary to qualify the issuance of the Closing Shares for such exemption.
- 2.7 **Sales, Use and Other Taxes.** The Company shall be responsible for all sales, use, documentary stamp and other Taxes, if any, arising out of the sale of the Purchased Assets to Purchaser pursuant to this Agreement or any of the transactions contemplated by this Agreement.
- 2.8 **Bulk Sales Compliance.** The Purchaser hereby waives compliance by the Company with the provisions of any and all laws relating to bulk transfers in connection with the sale of the Purchased Assets. The Company covenants and agrees to indemnify and hold harmless Purchaser from and against any and all Damages arising out of noncompliance with such bulk transfers laws.

Chksum: 1025560 Cycle 1

2.9 Closing.

- (a) Time and Place. The consummation of the purchase and sale of the Purchased Assets under this Agreement (the "Closing") shall take place at the offices of CIE, 2901 Business Center Drive, Suite 100, Irvine, CA, at 10:00 a.m. on January , 2004 or at such other time and in such manner as the parties mutually agree in writing (the "Closing Date").
- (b) Closing Deliveries by the Company. At the Closing, the Company shall have delivered or caused to be delivered to Purchaser:
 - possession of all of the Purchased Assets; (i)
 - (ii) the License Agreement, duly executed by the Company;
- (iii) a Bill of Sale substantially in the form of Exhibit B attached hereto, conveying good and marketable title in and to all of the Purchased Assets, duly executed by the Company;
- an Assignment and Assumption Agreement substantially in the form of Exhibit C attached hereto (the "General Assignment"), duly executed by the Company;
 - the Registration Rights Agreement, duly executed by the Company; (v)
 - (vi) the Updated Company Disclosure Schedule;
 - (vii) a certificate of an officer of the Company with respect to the matters set forth in Section 7.1

hereof:

- a certificate of the Secretary of the Company, certifying as of the Closing Date (A) a true and (viii) complete copy of the Certificate of Incorporation of the Company, (B) a true and complete copy of the resolutions of each of the board of directors and stockholders of the Company authorizing the execution, delivery and performance of this Agreement by the Company and the consummation of the transaction contemplated hereby, (C) a certificate of good standing as of a recent date of the Company in the State of California and (D) incumbency matters; and
- (ix) consents to assignment of the Assumed Contracts in form and substance satisfactory to Purchaser and its counsel.
 - (c) <u>Closing Deliveries By Purchaser</u>. At the Closing, Purchaser shall have delivered or caused to be delivered:
 - (i) the Closing Shares to the Company;
 - (ii) the License Agreement, duly executed by Purchaser;
 - (iii) the General Assignment, duly executed by Purchaser;
 - (iv) the Registration Rights Agreement, duly executed by Purchaser;
 - (v) the Updated Capitalization Representation, duly prepared by Purchaser;
 - the Updated Company Disclosure Schedule; (vi)
 - (vii) a certificate of an officer of Purchaser with respect to the matters set forth in Section 6.1 hereof;
- a certificate of the Secretary of Purchaser, certifying as of the Closing Date (A) a true and (viii) complete copy of the Certificate of Incorporation of the Purchaser, (B) a true and complete copy of the resolutions of the board of directors of the Purchaser authorizing the execution, delivery and performance of this Agreement by the Purchaser and the consummation of the transactions

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 19431 Cycle 1 EX-2.1 Purchasesoft, Inc.

contemplated hereby, (C) a certificate of good standing as of a recent date of the Purchaser in the State of Delaware and (D) incumbency matters; and

(ix) such other documents as the Company may reasonably request for the purpose of facilitating the consummation of the transactions contemplated herein.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Purchaser as of the date hereof, except as set forth on the Disclosure Schedule furnished separately to Purchaser or as set forth in the Company Reports, which exceptions shall be deemed to be representations and warranties as if made hereunder, as follows:

- 3.1 **Organization of the Company.** The Company is a corporation duly formed and validly existing under the laws of its jurisdiction of organization and has full corporate power and authority and legal right to own and operate the Purchased Assets and to carry on the Business as presently conducted, to execute and deliver this Agreement and all of the other agreements and instruments to be executed and delivered by the Company pursuant hereto, and to consummate the transactions contemplated hereby and thereby.
- 3.2 **Authority.** The Company has all necessary corporate power and corporate authority and has taken all action necessary to enter into this Agreement, to consummate the transactions contemplated hereby and to perform its obligations hereunder and no other proceedings on the part of the Company are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Company and constitutes a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms except (i) as limited by applicable bankruptcy, insolvency, reorganization, moratorium and other laws of general application affecting enforcement of creditors' rights generally and (ii) as limited by laws relating to the availability of specific performance, injunctive relief or other equitable remedies.
- 3.3 Financial Statements and Reports. To the Company's actual knowledge, the reports and documents provided by the Company (including without limitation Schedule 2.6) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. The unaudited financial statements and unaudited interim financial statements of Company included in the Company Reports (i) complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto when filed, (ii) were prepared in accordance with GAAP applied on a consistent basis throughout the periods covered thereby, (iii) fairly present the consolidated financial condition, results of operations and cash flows of Company as of the respective dates thereof and for the periods referred to therein, and (iv) are consistent with the books and records of Company. Since the date the reports have been provided, there has been no material adverse change in the financial condition or results operations of Company that has resulted in a Material Adverse Effect on the Business or the Purchased Assets. There are no restatements of Company's financial statements currently contemplated.
- (a) **No Conflicts.** The execution and delivery by the Company of this Agreement does not, and the performance by the Company of its obligations under this Agreement and the consummation of the transactions contemplated hereby will not:
- (b) conflict with or result in a violation or breach of any of the terms, conditions or provisions of the Certificate of Incorporation or Bylaws of the Company;

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 273398 Cycle 1 EX-2.1 Purchasesoft, Inc.

(c) conflict with or result in a violation or breach of any term or provision of any law, Order, Permit, statute, rule or regulation applicable to the Business or the Purchased Assets;

- (d) result in a breach of, or default under (or give rise to right of termination, cancellation or acceleration under) any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license, permit, agreement, lease or other similar instrument or obligation to which the Business or the Purchased Assets may be bound; or
 - (e) result in an imposition or creation of any Encumbrance on the Business or any of the Purchased Assets.
- 3.4 No Conflicts. The execution and delivery by the Company of this Agreement does not, and the performance by the Company of its obligations under this Agreement and the consummation of the transactions contemplated hereby will not:
- (a) conflict with or result in a violation or breach of any of the terms, conditions or provisions of the Certificate of Incorporation or Bylaws of the Company;
- (b) conflict with or result in a violation or breach of any term or provision of any law, Order, Permit, statute, rule or regulation applicable to the Business or the Purchased Assets;
- (c) result in a breach of, or default under (or give rise to right of termination, cancellation or acceleration under) any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license, permit, agreement, lease or other similar instrument or obligation to which the Business or the Purchased Assets may be bound; or
 - (d) result in an imposition or creation of any Encumbrance on the Business or any of the Purchased Assets.
- 3.5 **Consents, Approvals and Filings.** Other than with respect to Wells Fargo Business Line of Credit (account # 5474-6420-300-18189), no consent, approval or action of, filing with or notice to any Governmental or Regulatory Authority or other Persons on the part of the Company is required in connection with the execution, delivery and performance of this Agreement or the consummation of the transactions contemplated hereby other than approval by the Company's stockholders in accordance with the Company's bylaws and the California General Corporation Law.
- 3.6 **No Undisclosed Liabilities.** Other than certain prepaid software maintenance obligations, to the Company's actual knowledge, except as disclosed in the Company Reports (including financial statements contained therein), there are no material Liabilities, nor any basis for any claim against the Company for any such material Liabilities, relating to or affecting the Business or the Purchased Assets, other than Liabilities incurred after the date of the latest Company Report in the Ordinary Course of Business which have not had, and could not reasonably be expected to result in, individually or in the aggregate, a Material Adverse Effect on the Business or the Purchased Assets.
- 3.7 **Purchased Assets.** The location of all of the Purchased Assets, as of the Closing Date, shall be 2091 Business Center Drive, Suite 100 Irvine, CA 92612. The Company has good and marketable title to, or a valid leasehold interest in all of the Purchased Assets, free and clear of all Encumbrances (other than Permitted Encumbrances), except as set forth in Schedule 2.1(d). The Purchased Assets (together with the Excluded Assets), constitute all property of any nature owned by the Company used in, or useful to, the operation of the Business as conducted as of the date hereof. All tangible personal property of the Company and/or used in or useful to the operation of the Business, and included within the definition of Purchased Assets is in good operating condition and repair, ordinary wear and tear excepted. The Company shall be in actual possession of all of the Purchased Assets at the Closing.
- 3.8 **Benefit Plans; ERISA.** Except as set forth on Section 3.8 of the Company Disclosure Schedule, Purchaser will incur no liability with respect to, or on account of, and the Company will retain

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

Chksum: 101879 Cycle 1

Doc 2 Page 15

any liability for, and on account of, any employee benefit plan of the Company, including, but not limited to, liabilities the Company may have to such employees under all incentive compensation plans, bonus plans, pension and retirement plans, profit-sharing plans (including, any profit-sharing plan with a cash-or-deferred arrangement subject to Section 401(k) of the Code) stock purchase and option plans, savings and similar plans, medical, dental, travel, accident, life, disability and other insurance and other plans or arrangements, whether written or oral and whether "qualified" or "non-qualified" under the Code, or to any employee as a result of termination of employment by the Company as contemplated by this Agreement. The Company has not, with respect to any Offer Recipients, maintained, contributed to, or been obligated or required to contribute to, a "multiemployer plan," as such term is defined in Section 3(37) of ERISA. The Company is not a party to any collective bargaining agreement covering any Offer Recipients and the Company knows of no effort to organize any such employee as a part of any collective bargaining unit.

3.9 Real Property. The Company is not selling any real property. Suite 100 at 2091 Business Center Drive, Irvine, CA 92612 is the only parcel of real property leased by the Company or any of its Affiliates in the conduct of the Business (as lessee or lessor) (the "Leased Real Property"). To the Company's actual knowledge the Company has a valid leasehold interest in all real property used in or relating to the conduct of the Business, free and clear of all Encumbrances (other than Permitted Encumbrances). Each lease with respect to the Leased Real Property is a legal, valid and binding agreement subsisting in full force and effect enforceable in accordance with its terms, there is no, and the Company has not received notice of any, default (or any condition or event which, after notice or lapse of time or both, would constitute a default) thereunder. Such leases in effect allow the particular use of the premises involved, and no provision of any lease prohibits or unduly limits the Company's ability to conduct the Business so as to have a Material Adverse Effect on the Business if enforced. The Company does not owe any brokerage commissions with respect to any such Leased Real Property.

3.10 Intellectual Property Rights.

- (a) The Company has no patents, trademarks and copyright registrations and applications comprising Company's Intellectual Property.
- (b) There are no Actions or Proceedings before any Governmental or Regulatory Authority (including the United States Patent and Trademark Office or equivalent authority anywhere in the world) related to any Company Intellectual Property. No Company Intellectual Property or product or service of the Business is subject to any proceeding or outstanding decree, order, judgment, agreement, or stipulation restricting in any manner the use, transfer, or licensing thereof by the Company, or which may affect the validity, use or enforceability of such Company Intellectual Property.
 - (c) (Intentionally deleted)
- (d) To the extent that any Company Intellectual Property has been developed or created by a third party for the Company, the Company has a agreement with such third party with respect thereto and the Company thereby has obtained ownership of, and is the exclusive owner of, or has a valid license to use, all Intellectual Property in such work, material or invention by operation of law or by valid assignment.
- (e) The Company has not transferred ownership of, or granted any license (exclusive or non-exclusive) with respect to any Company Intellectual Property to any third party.
- (f) Section 3.10(f) lists all contracts, licenses and agreements to which the Company is a party that are currently in effect (i) with respect to the Company Intellectual Property licensed or offered to any third party; or (ii) pursuant to which a third party has licensed or transferred any Company Intellectual Property to the Company. Each of the contracts, licenses and agreements listed in Section 3.10(f) of the Company Disclosure Schedule is in full force and effect. Except for the licenses

Merrill Corp	poration -04-2032-1 Sat Feb 07 01:03:32 2004	Chksum: 884586 Cycle
EX-2.1	Purchasesoft, Inc.	

related to Lead Tools development products, which is nontransferable. The consummation of the transactions contemplated by this Agreement will neither violate nor result in the breach, modification, cancellation, termination or suspension of such contracts, licenses and agreements. The Company is in compliance with, and has not breached any term of any of such contracts, licenses and agreements. To the Knowledge of the Company, following the Closing Date Purchaser will be permitted to exercise all of the Company's rights under the contracts, licenses and agreements required to be listed in Section 3.10(f) to the same extent the Company would have been able to had the transactions contemplated by this Agreement not occurred and without the payment of any additional amounts or consideration other than ongoing fees, royalties or payments which the Company would otherwise be required to pay.

- (g) To the actual Knowledge of the Company, the operation of the Business, as currently conducted, has not, does not and will not infringe or misappropriate the Intellectual Property of any third party or constitute unfair competition or trade practices under the laws of any jurisdiction.
- (h) The Company has not received notice from any third party that the operation of the Business or any act, product or service of the Business infringes or misappropriates the Intellectual Property of any third party or constitutes unfair competition or trade practices under the laws of any jurisdiction.
- (i) To the actual Knowledge of the Company, (i) no Person has or is infringing or misappropriating any Company Intellectual Property and (ii) there have been, and are, no claims asserted against the Company or against any customer of the Company, related to any product or service of the Business.
- 3.11 **Litigation.** To the Company's actual knowledge, there are no Actions or Proceedings pending or, to the actual Knowledge of the Company, threatened or anticipated against, relating to or affecting (i) the Business or the Purchased Assets or (ii) the transactions contemplated by this Agreement, and, to the Knowledge of the Company, there is no basis for any such Action or Proceeding. The Company is not in default with respect to any Order, and there are no unsatisfied judgments against the Company.
- 3.12 **Compliance with Law.** To the actual Knowledge of the Company, it is in compliance with all applicable laws, statutes, Orders, ordinances and regulations, whether federal, state, local or foreign, except where the failure to comply, in each instance and in the aggregate, would not reasonably be expected to have a Material Adverse Effect on the Business. The Company has not received any notice to the effect that, or otherwise has been advised that, it is not in compliance with any of such laws, statutes, Orders, ordinances or regulations, where the failure to comply could reasonably be expected to result in a Material Adverse Effect on the Business.

3.13 Contracts.

- (a) To the best of the Company's knowledge, Section 3.13 of the Disclosure Schedule contains a true and complete list of each of the following contracts, agreements or other arrangements to which the Company is a party and by which any of the Purchased Assets are bound:
- (i) all loan agreements, indentures, debentures, notes or letters of credit relating to the borrowing of money or to mortgaging, pledging or otherwise placing an Encumbrance on any Purchased Assets;
- (ii) all leases or agreements under which the Company is lessee or lessor of, or holds, or operates, any property, real or personal, owned by any other party used in connection with the conduct of the Business;

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 336216 Cycle 1 EX-2.1 Purchasesoft, Inc.

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

(iii) all commitments, contracts, sales contracts, purchase orders, mortgage agreements or groups of related agreements with the same party or any group or affiliated parties which require or may in the future require payment of any consideration by the Company;

- (iv) all license agreements, distribution agreements or any other agreements involving any of the Company Intellectual Property;
 - (v) all contracts or commitments that in any way restrict the Company from carrying on the Business;

Doc 2 Page 17

- (vi) all other contracts and agreements pertaining to the conduct of the Business or by which any of the Purchased Assets is bound that (A) involve the payment or potential payment, pursuant to the terms of any such contract or agreement, by the Company and (B) cannot be terminated within thirty (30) days after giving notice of termination without resulting in any cost or penalty to the Company; and
- (vii) all contracts or commitments that in any way grants a third party a right of first refusal for the purchase of any of the Purchased Assets.
- (b) A correct and complete copy of each contract, agreement or other arrangement disclosed in Section 3.13 of the Company Disclosure Schedule has been previously provided to Purchaser. Each contract, agreement or other arrangement disclosed in Section 3.13 of the Company Disclosure Schedule is in full force and effect and to the Company's actual knowledge constitutes a legal, valid and binding agreement, enforceable in accordance with its terms.
- 3.14 Environmental Matters. To the Company's actual knowledge the Business is in compliance with, and has at all times complied with, all applicable Environmental Laws in all respects and, to the Knowledge of the Company, there are no circumstances which may prevent or interfere with such compliance in the immediate future. In the last five (5) years, the Company has not received any communication (whether written or oral), whether from a Governmental or Regulatory Authority, citizen group, employee or otherwise, that alleges that the Company or any predecessor of any of the Leased Real Property or Purchased Assets is not in full compliance with Environmental Laws. All Permits, licenses, registrations and other governmental authorizations currently held by the Company pursuant to Environmental Laws are identified in Section 3.14 of the Company Disclosure Schedule and are in good standing and without any violation and to the Company's actual knowledge represent all such environmental Permits necessary for the conduct of the Business as currently conducted. The Company has not been notified by any Governmental or Regulatory Authority that any environmental Permit will be modified, suspended or revoked or cannot be renewed, reissued or transferred, and, to the actual Knowledge of the Company, no environmental Permit will be modified, suspended or revoked, or cannot be renewed, reissued or transferred.
- 3.15 **Inventory.** The inventory of the Business is in good and merchantable condition, and suitable and usable at its carrying value in the Ordinary Course of Business for the purposes for which intended. There is no material adverse condition affecting the supply of materials available to the Company in the conduct of the Business. All inventories used in or relating to the conduct of the Business are owned by the Company free and clear of any Encumbrances (other than Permitted Encumbrances).
- 3.16 **Plants, Buildings, Structures, Facilities and Equipment.** To the Company's actual knowledge, all plants, buildings, structures, facilities and equipment used by the Company in the conduct of the Business are structurally sound with no known material defects and are in good operating condition and repair (subject to normal wear and tear) so as to permit the operation of the Business as presently conducted.

 Merrill Corporation -04-2032-1
 Sat Feb 07 01:03:32 2004
 Chksum: 142028 Cycle 1

 EX-2.1
 Purchasesoft, Inc.

 C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf
 EDGAR
 Doc 2 Page 18

3.17 **Customer Lists and Accounts.** The Customer Lists and Accounts set forth in <u>Schedule 2.1(b)</u> is a true and correct list of the Company's customers and accounts as of the date hereof.

- 3.18 **Relationships with Suppliers and Licensors.** No current supplier to the Company with the respect to the Business has notified the Company of an intention to terminate or substantially alter its existing business relationship with the Company nor has any licensor under a license agreement with the Company that constitutes part of the Assumed Contracts notified the Company of an intention to terminate or substantially alter the Company's rights under such license.
- 3.19 **Tax Matters.** All Taxes of the Company against the Purchased Assets have been or will be paid on a timely basis. The Company has duly and timely filed (or will file prior to the Closing) all Tax Returns required to be filed prior to Closing, and all such Tax Returns and reports are true, correct, and complete in all material respects. There are no Encumbrances (other than Permitted Encumbrances) for Taxes on any of the Purchased Assets. The Company has complied with all record keeping and tax reporting obligations relating to income and employment taxes due with respect to compensation paid to employees or independent contractors. The Company is not a "foreign person" within the meaning of Section 1445(f)(3) of the Code. There are no pending or, to the Knowledge of the Company, threatened proceedings with respect to Taxes for which Purchaser could bear successor liability beyond what is set forth in the Disclosure Schedule or which could become a charge against the Purchased Assets, and there are no outstanding waivers or extensions of statutes of limitations with respect to assessments of Taxes, of the Company for which Purchaser could bear successor liability beyond what is set forth in the Disclosure Schedule or which could become a charge against the Purchased Assets.
- 3.20 **Permits.** Section 3.20 of the Disclosure Schedule contains a true and complete list of all Permits used by the Company in the conduct of the Business. All such Permits are currently effective and valid and have been validly issued and are not freely transferable to Purchaser at the Closing. To the actual Knowledge of the Company, there is no pending Action or Proceeding by any Governmental or Regulatory Authority which could affect the Permits or their sufficiency for the current conduct of the Business or of the conduct of the Business after the Closing.
- 3.21 **Brokers.** The Company has not retained any broker in connection with the transactions contemplated hereunder. Purchaser has, and will have, no obligation to pay any broker's, finder's, investment banker's, financial advisor's or similar fee in connection with this Agreement or the transactions contemplated hereby by reason of any action taken by or on behalf of the Company.
- 3.22 **Material Misstatements and Omissions.** The statements, representations and warranties of the Company contained in this Agreement and in each document, statement, certificate or exhibit furnished or to be furnished by or on behalf of the Company pursuant hereto, or in connection with the transactions contemplated hereby, taken together, do not contain and will not contain any untrue statement of a material fact and do not or will not omit to state a material fact necessary to make the statements or facts contained herein or therein, in light of the circumstances made, not misleading.
- 3.23 **Investment Representations.** The Company is an "accredited investor" as such term is defined in Rule 501(a) promulgated under the Securities Act. The Company is aware that the Closing Shares have not been registered under the Securities Act or any applicable state securities laws, and hereby agrees that the Closing Shares may not be offered or sold (i) in the absence of registration under the Securities Act and any applicable state securities laws or an exemption from the registration requirements of the Securities Act and any applicable state securities laws and (ii) unless in compliance with the terms and provisions of this Agreement. The Company represents that the Company is familiar with Rule 144 promulgated by the SEC pursuant to the Securities Act, as presently in effect, and understands the resale limitations imposed thereby and by the Securities Act. The Company understands that the offering and sale of the Closing Shares is intended to be exempt from registration under the Securities Act, by virtue of the private placement exemption provided by Rule 505 and/or 506 of

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 660722 Cycle 1 EX-2.1 Purchasesoft, Inc.

Regulation D promulgated under the Securities Act and/or Section 4(2) of the Securities Act, based, in part, upon the representations, warranties and agreements contained in this Agreement, and Purchaser may rely on such representations, warranties and agreements in connection therewith.

The Company agrees that it will be acquiring the Closing Shares for its own account and for investment, and not with a view to the distribution thereof or with any present intention of distributing or selling any of the Closing Shares except in compliance with the Securities Act, applicable state securities laws and this Agreement. The Company represents that by reason of its business and financial experience, the Company has knowledge, sophistication and experience in business and financial matters as to be capable of evaluating the merits and risk of the prospective investment. The financial condition and investments of the Company are such that the Company is in a financial position to hold the Closing Shares for an indefinite period of time and to bear the economic risk of, and withstand a complete loss of, the investment in the Closing Shares.

3.24 <u>The Company's Actual Knowledge Defined</u>. For purposes of this Agreement, the actual knowledge shall mean the actual knowledge of Craig Chandler or Kathy A. Chandler, without duty of further inquiry.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser represents and warrants to the Company as of the date hereof, except as set forth on the Purchaser Disclosure Schedule furnished separately to the Company, which exceptions shall be deemed to be representations and warranties as if made hereunder, as follows:

- 4.1 **Organization.** Purchaser is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware. Purchaser has the requisite corporate power and corporate authority to carry on its business and own its Assets and Properties except where failure to have such power and authority would not have a Material Adverse Effect on Purchaser. Purchaser is duly qualified to conduct its respective business and is in good standing under the laws of each jurisdiction where such qualification is required except for any jurisdiction where failure so to qualify would not have a Material Adverse Effect upon Purchaser.
- 4.2 Capital Stock of Purchaser. The authorized capital stock of Purchaser, consists of (i) 50,000,000 shares of common stock, par value \$0.01 per share, of which 33,287,267 shares were issued and outstanding as of the date hereof; (ii) no shares of capital stock of Purchaser in treasury; and (iii) no shares of Preferred Stock, authorized, issued or outstanding. Each share of the issued and outstanding capital stock of Purchaser is duly authorized, validly issued, fully paid and nonassessable. Purchaser will provide at Closing an updated version of this Section 4.2, updated and accurate as of the Closing (the "Updated Capitalization Representation"). The fully diluted percentage ownership of Purchaser represented by the Closing Shares will not materially change between the date hereof and the Closing.

Except for (i) outstanding warrants to purchase 375,000 shares of common at \$0.01 per share that were issued in connection with bridge loans, (ii) outstanding warrants to purchase 520,000 shares of common at \$0.02 per share that were issued in connection with distribution agreements and services rendered, and (iii) outstanding agreements to issue 11,800,000 shares that may be issued in connection with pending acquisitions, there are no outstanding options, warrants, rights (including conversion or preemptive rights) or agreements for the purchase or acquisition from Purchaser of any shares of its capital stock. In addition, Purchaser intends to reserve 1,000,000 shares of its Common Stock for purchase upon exercise of options to be granted in the future under the Company's option plan.

 Merrill Corporation -04-2032-1
 Sat Feb 07 01:03:32 2004
 Chksum: 916368 Cycle 1

 EX-2.1
 Purchasesoft, Inc.

 C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf
 EDGAR
 Doc 2 Page 20

- 4.3 **Authority.** Purchaser has all necessary corporate power and corporate authority and has taken all corporate action necessary to enter into this Agreement, to consummate the transactions contemplated hereby and to perform its respective obligations hereunder and no other proceedings or corporate or stockholder action on the part of Purchaser is necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Purchaser and (assuming due authorization, execution and delivery by the other parties to this Agreement) constitutes a legal, valid and binding obligation of Purchaser enforceable against Purchaser in accordance with its terms except (i) as limited by applicable bankruptcy, insolvency, reorganization, moratorium and other laws of general application affecting enforcement of creditors' rights generally and (ii) as limited by laws relating to the availability of specific performance, injunctive relief or other equitable remedies.
- 4.4 **Litigation.** There are no Actions or Proceedings pending or, to the Knowledge of Purchaser, threatened against Purchaser relating to or affecting (i) Purchaser, or its Assets and Properties or its businesses, or (ii) the transactions contemplated by this Agreement, and, to the Knowledge of Purchaser, there are no facts or circumstances that could reasonably be expected to give rise to any such Action or Proceeding. Purchaser is not in default with respect to any Order, and there are no unsatisfied judgments against Purchaser. Purchaser has not received notice, or otherwise has no knowledge of any Orders outstanding against Purchaser.
- 4.5 Financial Statements and reports. All reports and documents filed by Purchaser under Section 13 or subsections (a) or (c) of Section 14 of the Exchange Act with the SEC (such reports are collectively referred to herein as the "Purchaser Reports") constitute all of the reports and documents required to be filed by Purchaser under Section 13 or subsections (a) or (c) of Section 14 of the Exchange Act with the SEC from August 5th, 2002 through January 1, 2004. The Purchaser Reports have been duly and timely filed, were in compliance in all material respects with the requirements of the Exchange Act and the rules and regulations thereunder when filed, and were complete and correct in all material respects as of the dates at which the information therein was furnished. As of their respective dates, the Purchaser Reports did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. The audited financial statements and unaudited interim financial statements of Purchaser included in the Purchaser Reports (i) complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto when filed, (ii) were prepared in accordance with GAAP applied on a consistent basis throughout the periods covered thereby (except as may be indicated therein or in the notes thereto, and in the case of quarterly financial statements, as permitted by Form 10-QSB under the Exchange Act), (iii) fairly present the consolidated financial condition, results of operations and cash flows of Purchaser as of the respective dates thereof and for the periods referred to therein, and (iv) are consistent with the books and records of Purchaser. Since the date of the filing with the SEC of Purchaser's most recent Form 10-QSB, there has been no material adverse change in the financial condition or results operations of Purchaser that has resulted in a Material Adverse Effect on the Business or the Purchased Assets. There are no restatements of Purchaser's financial statements currently contemplated as discussed with Purchaser's Audit Committee. Purchaser acknowledges that it has access to, and has reviewed (to the extent it has deemed necessary) the Company Reports.
- 4.6 **Financial Statements.** Purchaser has delivered to the Company its audited balance sheet and audited statements of income and cash flows for the fiscal year ending May 31, 2003, and its unaudited balance sheet and unaudited statements of income and cash flows for the three month period ending August 31, 2003 (the "Statement Date") (all of the foregoing financial statements, collectively, the "Financial Statements"). The Financial Statements, together with the notes thereto, are complete and correct in all material respects, have been prepared in accordance with generally accepted accounting principles applied on a consistent basis throughout the periods indicated, except as disclosed therein, and

 Merrill Corporation -04-2032-1
 Sat Feb 07 01:03:32 2004
 Chksum: 292247 Cycle 1

 EX-2.1
 Purchasesoft, Inc.

 C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf
 EDGAR
 Doc 2 Page 21

present fairly the financial condition and position of Purchaser as of the dates, and for the periods, specified therein; provided, however, that the unaudited Financial Statements are subject to normal recurring year-end audit adjustments (which are not expected to be material), and do not contain all footnotes required under generally accepted accounting principles.

- 4.7 **No Undisclosed Liabilities.** Purchaser has no material liabilities and knows of no material contingent liabilities not disclosed in the Financial Statements, except current liabilities incurred in the ordinary course of business subsequent to the Statement Date which have not been, either in any individual case or in the aggregate, materially adverse.
- 4.8 **Brokers.** Purchaser has not retained any broker in connection with the transactions contemplated hereunder. Purchaser will have no obligation to pay any broker's, finder's investment banker's, financial advisor's or similar fee in connection with this Agreement or the transactions contemplated hereby.
- 4.9 **No Conflicts.** The execution and delivery by Purchaser of this Agreement does not, and the performance by Purchaser of its respective obligations under this Agreement and the consummation of the transactions contemplated hereby will not:
- (a) conflict with or result in a violation or breach of any of the terms, conditions or provisions of the Certificate of Incorporation or Bylaws of Purchaser;
- (b) conflict with or result in a violation or breach of any term or provision of any law, Order, Permit, statute, rule or regulation of a Governmental or Regulatory Authority applicable to Purchaser, the business or Assets or Properties of Purchaser or the capital stock of Purchaser; or
- (c) result in a breach of, or default under (or give rise to a right of termination, cancellation or acceleration) under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license, agreement, lease or other similar instrument or obligation to which Purchaser may be bound; or
- (d) result in an imposition or creation of any Encumbrance (other than a Permitted Encumbrance) on the business or Assets or Properties of Purchaser.
- 4.10 **Consents and Governmental Approvals and Filings.** No consent, approval or other action of, filing with or notice to any Governmental or Regulatory Authority on the part of Purchaser is required in connection with the execution, delivery and performance of this Agreement or the consummation of the transactions contemplated hereby.
- 4.11 **Compliance with Law.** To the Knowledge of Purchaser, it is in compliance with all applicable laws, statutes, Orders, ordinances and regulations, whether federal, state, local or foreign, except where the failure to comply, in each instance and in the aggregate, would not reasonably be expected to have a Material Adverse Effect on Purchaser. Purchaser has not received any notice to the effect that, or otherwise has been advised that, it is not in compliance with any of such laws, statutes, Orders, ordinances or regulations, where the failure to comply could reasonably be expected to result in a Material Adverse Effect on Purchaser.
 - 4.12 Ownership of Purchaser; No Prior Activities. Not applicable.
- 4.13 **Trading of Securities.** Neither Purchaser nor any of its officers, directors, shareholders, employees, affiliates, agents or representatives have violated any laws, statutes, Orders, ordinances and regulations, whether federal, state, local or foreign, arising out of or in any way related to the issuance of or trading in the capital securities of Purchaser, including, without limitation, Rule 10b-5 of the Exchange Act.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 8702 Cycle 1

EX-2.1 Purchasesoft, Inc.

4.14 **Closing Shares.** The Closing Shares, upon issuance thereof, will be duly authorized, validly issued, fully paid, nonassessable, and not subject to any Encumbrance. The Closing Shares shall be issued in compliance with all applicable securities laws.

4.15 **Material Misstatements and Omissions.** The statements, representations and warranties of Purchaser contained in this Agreement (including the exhibits and schedules hereto) and in each document, statement, certificate or exhibit furnished or to be furnished by or on behalf of Purchaser pursuant hereto, or in connection with the transactions contemplated hereby, taken together, do not contain and will not contain any untrue statement of a material fact and do not or will not omit to state a material fact necessary to make the statements or facts contained herein or therein, in light of the circumstances made, not misleading.

ARTICLE VCOVENANTS OF THE PARTIES

Each of the parties covenants with the others to act, as follows:

- 5.1 Operation of Business Prior to Closing Date. Except as otherwise contemplated by this Agreement, between the date hereof and the Closing Date (or earlier termination of this Agreement), the Company will operate the Business in the Ordinary Course of Business and, to the extent consistent therewith, with no less diligence and effort than would be applied in the absence of this Agreement, use all commercially reasonable efforts to seek to preserve intact its current Business organizations, keep available the service of current managers, officers and employees of the Business and preserve relationships with customers, suppliers, distributors, lessors, employees, contractors and others having business dealings with the Business with the intention that the Business shall be unimpaired at the Closing Date. Without limiting the generality of the foregoing, except as otherwise expressly provided in this Agreement, prior to the Closing Date (or earlier termination of this Agreement), the Company:
- (a) will not create, incur or assume any obligation which would adversely affect the Purchased Assets or Purchaser's ability to conduct the Business in substantially the same manner and condition as conducted by the Company on the date of this Agreement;
- (b) except to the extent an obligation is established in a written agreement in existence prior to the date hereof, will not change in any manner the compensation of, or agree to provide additional benefits to, or enter into any employment agreement with, any Offer Recipient, except as contemplated in Section 5.9 below and except for such benefits provided to substantially all of the Company's similarly situated employees;
 - (c) will maintain insurance coverage in amounts adequate to cover the reasonably anticipated risks of the Business;
- (d) will not sell, dispose of or encumber any of the Purchased Assets or license any Purchased Assets to any Person except object code licenses on a non-exclusive basis in a manner and on terms consistent with past practice;
- (e) will not enter into any material agreements or commitments relating to the Business, except on commercially reasonable terms in the Ordinary Course of Business of the Business;
 - (f) will comply in all material respects with all laws and regulations applicable to the Business;
 - (g) will not enter into any agreement with any third party for the distribution of any of the Purchased Assets;
- (h) will not materially amend its Certificate of Incorporation or Bylaws (except as necessary to comply with the terms of this Agreement)

 Merrill Corporation -04-2032-1
 Sat Feb 07 01:03:32 2004
 Chksum: 179060 Cycle 1

 EX-2.1
 Purchasesoft, Inc.

 C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf
 EDGAR
 Doc 2 Page 23

5.2 **Investigation by Purchaser.** Subject to all applicable confidentiality laws and confidentiality obligations of the Company, the Company shall allow Purchaser or its authorized representatives, at Purchaser's own expense during regular business hours, or otherwise with the consent of the Company (which consent shall not be unreasonably withheld), to make such inspection of the Company and to inspect (and, if applicable, make copies of) Books and Records, plants, offices, warehouses and other facilities of the Company as reasonably requested by Purchaser or its authorized representatives.

- 5.3 **Consents.** As soon as practicable after execution of this Agreement, each party will commence all action required hereunder to obtain all applicable Permits, consents, approvals and agreements of, and to give all notices and make all filings with, any third parties as may be necessary to authorize, approve or permit the full and complete consummation of the transactions contemplated hereby by the Closing Date.
- 5.4 **Notification of Certain Matters.** Each of the parties shall give prompt notice to the other party, of (i) the discovery of a fact or facts of which the notifying party has actual knowledge which cause it to conclude that any of the representations, warranties or statements made by it or in an any exhibit, schedule or other document delivered pursuant to this Agreement, may be false or misleading or omission of any facts necessary in order to make such representations, warranties or statements not false or misleading; (ii) the occurrence, or failure to occur, of any event which occurrence or failure would be likely to cause any representation or warranty made by them in this Agreement to be untrue or inaccurate any time from the date hereof to the Closing Date; and (iii) any failure of the notifying party to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder. Each party hereto shall use all reasonable efforts to remedy any failure on its part to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder.
- 5.5 Cooperative Efforts. Subject to the terms and conditions of this Agreement, each of the parties hereto will use its commercially reasonable efforts to take, or cause to be taken, all action, or to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to consummate and make effective the transactions contemplated by this Agreement, including, without limitation, obtaining all consents and approvals of all Persons and Governmental or Regulatory Authorities and removing any injunctions or other impairments or delays or otherwise which are necessary to the consummation of the transactions contemplated by this Agreement.
- 5.6 Filings. Each of the parties hereto will use its best efforts to make or cause to be made all such filings and submissions as may be required under applicable laws and regulations for the consummation of the transactions contemplated by this Agreement. The Company and Purchaser will coordinate and cooperate with one another in exchanging such information and provide each other such assistance as any other party may reasonably request in connection with the foregoing.
- 5.7 **Inconsistent Activities**. Unless and until this Agreement is terminated pursuant to Section 9.1, the Company will not directly or indirectly solicit, initiate or encourage any inquiries or proposals from, discuss or negotiate with, provide any non-public information to or consider the merits of any unsolicited inquiries or proposals from, any Person (other than Purchaser) relating to any transaction involving the sale of the Business or the Purchased Assets, or any merger, consolidation, business combination or similar transaction involving the Business (each a "Proposed Acquisition Transaction"). The Company will immediately notify Purchaser if any discussions or negotiations are sought to be initiated, any inquiry or proposal is made, or any information is requested with respect to any Proposed Acquisition Transaction and notify Purchaser of the terms of any proposal which it or its Affiliates, if any, may receive in respect of any such Proposed Acquisition Transaction, including without limitation the identity of the prospective purchaser or soliciting party.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 EX-2.1 Purchasesoft, Inc. C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

DGAR Doc 2 Page 24

Chksum: 883149 Cycle 1

5.8 **Public Announcements.** Except as may be required by applicable law, including any determination that a press release or other public statement or filing is required under applicable securities or regulatory rules, prior to the Closing, none of the parties hereto shall issue or cause the publication of any press release or otherwise make any public statement with respect to the transactions contemplated hereby without the prior written consent of each of the other parties hereto.

5.9 Employee Matters.

- (a) Offer of Employment. Subject to and in accordance with the provisions of this Section 5.9, Purchaser will offer employment to all employees other than , who are employed in the Business as of the date of this Agreement (such employees to receive offers of employment are referred to as the "Offer Recipients"). The Company has delivered to Purchaser a list setting forth the names, home addresses, compensation levels, stock option position, if any, and job titles of all Offer Recipients. Prior to the Closing, Purchaser, after notice to the Company as to the timing and method of contact, shall have the right to contact each of the Offer Recipients for the purposes of making offers of employment with Purchaser to be effective after the Closing Date and receiving written acceptances of such employment (in each case contingent on consummation of the transactions contemplated by this Agreement). Upon Closing, Purchaser shall hire all Offer Recipients who accept such offer in the manner and within the time frame reasonably established by Purchaser. Each such employee who is employed by the Company on the Closing Date and who actually transfers to employment with Purchaser at or after the Closing Date as a result of an offer of employment made by Purchaser is hereafter referred to as a "Transferred Employee." On a periodic basis following the date hereof and prior to the Closing, Purchaser shall advise the Company of its intentions with respect to Offer Recipients it desires to extend or has extended offers to and the general status of discussions with such employees.
- (b) <u>Transition</u>. The employment of the Transferred Employees by the Company shall end at the close of business on the Closing Date and the employment of the Transferred Employees by Purchaser shall commence at 12:01 a.m. on the day after the Closing Date, except as to those Transferred Employees who are on disability leave of less than twenty-six (26) weeks, authorized leave of absence or military service as of the Closing Date, in which case such Transferred Employees shall remain employees of the Company until, and will commence employment with Purchaser as of, 12:01 a.m. on the date they return to active employment. Transferred Employees shall not include any person on a disability leave of more than twenty-six (26) weeks. The terms of employment with Purchaser shall be as mutually agreed to between each Transferred Employee and Purchaser, subject to the succeeding provisions of this Section 5.9.
- (c) <u>Retention of Employees Prior to Closing</u>. The Company agrees to use reasonable efforts to (i) retain the Offer Recipients as employees of the Business until the Closing Date, and (ii) assist Purchaser in securing the employment after the Closing Date of the Offer Recipients. The Company shall notify Purchaser promptly if any Offer Recipient terminates employment with the Company after the date of this Agreement but prior to the Closing.
- (d) <u>Employees Other than Transferred Employees</u>. Any employees of the Company who do not become Transferred Employees will remain employees of the Company after the Closing. Any severance obligations to such employees shall be the Company's responsibility.
- 5.10 **Prorations.** The Purchaser and the Company agree to make customary prorations (as of the Closing Date) in respect of items customarily prorated in connection with the sale of assets similar to the Purchased Assets, including, without limitation, if applicable, real estate taxes and power and utility charges.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 134729 Cycle 1 EX-2.1 Purchasesoft, Inc.

5.11 **Confidentiality.** The terms of the Non Disclosure Agreement between the parties hereto, dated September 2, 2003 (the "Confidentiality Agreement"), a copy of which is attached hereto as <u>Exhibit E</u>, shall continue in full force and effect following the date hereof and the Closing Date.

- 5.12 **Approval of the Company's Stockholders.** If required by this transaction, the Company shall take the actions necessary to conduct a special meeting of the Company's stockholders to consider and vote on the transactions contemplated by this Agreement at the earliest practicable date after the date of this Agreement and in connection therewith the Company's Board of Directors shall recommend to the Company's Stockholders that they approve this Agreement and the transactions contemplated thereby. The Company agrees to use its commercially reasonable efforts to take all necessary steps to obtain approval of the Company's stockholders, including the filing and distribution of a proxy statement, calling of a special meeting and the holding of that meeting. Such a special meeting shall be called, held and conducted and proxies shall be solicited, in compliance with the Company's Certificate of Incorporation and Bylaws, both as amended, and in compliance with applicable law.
- 5.13 **Updating of Disclosure Schedules**. The Company shall prepare and deliver an updated version of the Company Disclosure Schedule (the "Updated Company Disclosure Schedule") and all schedules and exhibits thereto to include all information necessary to make the representations and warranties of the Company contained in this Agreement, as supplemented by the Updated Company Disclosure Schedule, accurate as of the Closing Date. The Company shall deliver a reasonably complete version of the Updated Company Disclosure Schedule to Purchaser approximately five (5) days prior to Closing. Purchaser shall prepare and deliver an updated version of the Purchaser Disclosure Schedule (the "Updated Purchaser Disclosure Schedule") and all schedules and exhibits thereto to include all information necessary to make the representations and warranties of Purchaser contained in this Agreement, as supplemented by the Updated Purchaser Disclosure Schedule, accurate as of the Closing Date. Purchaser shall deliver a reasonably complete version of the Updated Purchaser Disclosure Schedule to the Company approximately five (5) days prior to Closing.

ARTICLE VI CONDITIONS TO THE OBLIGATIONS OF THE COMPANY

The obligations of the Company to effect the transactions contemplated hereby are subject to the satisfaction, at or before the Closing, of each of the following conditions:

- 6.1 **No Material Adverse Effect.** Purchaser shall not have acted or caused any Person to have acted in any manner which has created or could reasonably be expected to create (individually or in the aggregate), a Material Adverse Effect on Purchaser.
- 6.2 **Stockholder Approval.** Purchaser shall have obtained the approval of its shareholders (if necessary) for the consummation of the transactions contemplated by this Agreement.
- 6.3 **Closing Deliveries.** Purchaser shall have executed and delivered the documents required to be executed and delivered by Purchaser pursuant to Section 2.9(c) above.

ARTICLE VII CONDITIONS TO THE OBLIGATIONS OF PURCHASER

The obligation of Purchaser to effect the transactions contemplated hereby is subject to the satisfaction, at or before the Closing, of each of the following conditions:

- 7.1 **Material Adverse Effect.** The Company shall not have acted or caused any Person to have acted in any manner which has created or could reasonably be expected to create (individually or in the aggregate), a Material Adverse Effect on the Business or the Purchased Assets.
- 7.2 **Stockholder Approval.** The Company shall have obtained the approval of its stockholders for the consummation of the transactions contemplated by this Agreement.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 527947 Cycle 1 EX-2.1 Purchasesoft, Inc.

7.3 **Closing Deliveries.** The Company shall have executed and delivered the documents required to be executed and delivered by the Company pursuant to Section 2.9(b) above.

ARTICLE VIII ACTIONS BY THE PARTIES AFTER THE CLOSING

8.1 Survival of Representations, Warranties, Etc. The representations, warranties and covenants contained in or made pursuant to this Agreement or any certificate, document or instrument delivered pursuant to or in connection with this Agreement in the transactions contemplated hereby shall survive the execution and delivery of this Agreement and the Closing hereunder (notwithstanding any investigation, analysis or evaluation by any party hereto or their designees of the Assets and Properties, business, operations or condition (financial or otherwise) of the other party), and thereafter the representations and warranties of the Parties herein shall continue to survive in full force and effect for a period of twelve (12) months after the Closing Date (the "Survival Period").

8.2 Indemnification.

- (a) By the Company. The Company shall indemnify, defend and hold harmless Purchaser and each of its officers, directors, employees, agents, successors and assigns (collectively the "Purchaser Group") from and against any and all costs, losses, Liabilities, damages, lawsuits, deficiencies, claims and expenses, including without limitation, penalties, costs of mitigation, clean-up or remedial action, reasonable attorneys' fees and all amounts paid to third parties in investigation, defense or settlement of any of the foregoing (collectively, the "Damages"), suffered by Purchaser, incurred in connection with, arising out of, resulting from or incident to (i) any breach of any covenant, representation, warranty or agreement or the inaccuracy of any representation, made by the Company in or pursuant to this Agreement and (ii) Liabilities that are not Assumed Liabilities.
- (b) <u>By Purchaser</u>. Purchaser shall indemnify, defend and hold harmless the Company, its officers, managers, employees, agents, successors and assigns (the "Company Group") from and against any and all Damages incurred in connection with, arising out of, resulting from or incident to any (i) breach of any covenant, representation, warranty or agreement or the inaccuracy of any representation made by Purchaser in or pursuant to this Agreement and (ii) Assumed Liabilities.
- (c) <u>Resolution of Claims</u>. A claim for indemnification for any matter not involving a third-party claim may be asserted by written notice to the party from whom indemnification is sought to the other party (the "Claim Notice"). Upon receipt of a Claim Notice, the party from whom indemnification is sought shall have fifteen (15) Business Days to object, in writing, to such claim (the "Dispute Notice"), otherwise the party seeking indemnification shall have the right to enforce its indemnity rights as defined hereunder. If the party from whom indemnification is sought provides the other party with a Dispute Notice in a timely fashion, the parties shall attempt in good faith to agree upon the rights or the respective parties with respect to such claim. If the parties agree as to the resolution of such claim, they shall prepare a memorandum setting forth the terms of such resolution signed by each of the parties hereto and enforce the indemnification rights hereunder. If no agreement is reached within thirty (30) days after delivery of the Dispute Notice, the dispute resolution provisions of this Agreement shall govern.
- (d) Third Party Claims; Defense of Claims. If any Action or Proceeding is filed or initiated against any party entitled to the benefit of indemnity hereunder, written notice thereof shall be given to the indemnifying party as promptly as practicable (and in any event within ten (10) days after the service of the citation or summons); provided, however, that a delay or failure of any indemnified party to give timely notice shall not affect rights to indemnification hereunder except to the extent that the indemnifying party demonstrates actual damage caused by such failure. Any such notice shall state (with reasonable specificity) the basis on which indemnification is being asserted, the amount of Damages for which indemnification is being asserted and copies of all relevant pleadings, demands and other papers

EDCAD

Doc 2 Page 27

Chksum: 31604 Cycle 1

being served on the indemnified party. After such notice, the indemnifying party may, if it so elects, take control of the defense and investigation of such Action or Proceeding and to employ and engage attorneys of its own choice to handle and defend the same, such attorneys to be reasonably satisfactory to the indemnified party, at the indemnifying party's sole cost, risk and expense (unless the indemnifying party has failed to assume the defense of such Action or Proceeding), and compromise or settle such Action or Proceeding, which compromise or settlement shall be made only with the written consent of the indemnified party, such consent not to be unreasonably withheld. The indemnified party may withhold such consent if such compromise or settlement would adversely affect the conduct of business or requires less than an unconditional release to be obtained. If the indemnifying party fails to assume the defense of such Action or Proceeding within fifteen (15) days after receipt of notice thereof pursuant to this Section 8.2, the indemnified party against which such Action or Proceeding has been filed or initiated will (upon delivering notice to such effect to the indemnifying party) have the right to undertake, at the indemnifying party's own cost and expense, the defense, compromise or settlement of such Action or Proceeding on behalf of and for the account and risk of the indemnifying party; provided, however, that such Action or Proceeding shall not be compromised or settled without the written consent of the indemnifying party, which consent shall not be unreasonably withheld. In the event the indemnified party assumes defense of the Action or Proceeding, the indemnified party will keep the indemnifying party reasonably informed of the progress of any such defense, compromise or settlement and will consult with, when appropriate, and consider any reasonable advice from, the indemnifying party of any such defense, compromise or settlement. The indemnifying party shall be liable for any settlement of any action subject to indemnification and effected pursuant to and in accordance with this Section 8.2 and for any final judgment (subject to any right of appeal), and the indemnifying party agrees to indemnify and hold harmless the indemnified party from and against any Damages by reason of such settlement or judgment.

The indemnified party shall cooperate in all reasonable respects with the indemnifying party and its attorneys in the investigation, trial and defense of such Action or Proceeding and any appeal arising therefrom; provided, however, that the indemnified party may, at its own cost, participate in the investigation, trial and defense of such Action or Proceeding and any appeal arising therefrom.

(e) Limitations on Indemnity.

The Company shall have no liability to Purchaser for amounts payable pursuant to its indemnification obligations in this Section 8.2 until the total of all such Damages incurred by any member of the Purchaser Group, individually or in the aggregate, exceed Fifty Thousand Dollars (\$50,000) (the "Threshold Amount"), and then indemnification by the indemnifying party shall apply to all such Damages exceeding the Threshold Amount. Purchaser shall have no liability to the Company for amounts payable pursuant to its indemnification obligations in this Section 8.2 until the total of all such Damages incurred by any member of the Company Group, individually or in the aggregate, exceed Fifty Thousand Dollars (\$50,000) (the "Threshold Amount"), and then indemnification by the indemnifying party shall apply to all such Damages exceeding the Threshold Amount.

The Indemnity Shares shall be the sole and exclusive means for Purchaser to collect any Damages for which it is entitled to indemnification under this Article VIII. The maximum aggregate amount of indemnification for any Damages for which the Company is required to indemnify the members of the Purchaser Group under this Agreement shall be limited to the aggregate value of the Indemnity Shares. The maximum aggregate amount of indemnification for any Damages for which Purchaser is required to indemnify the members of the Company Group under this Agreement shall be limited to the aggregate value of the Indemnity Shares.

The limitations on the Company's and Purchaser's indemnification obligations in 0 and 0 above shall not apply to any Damages arising out of or in connection with any fraud or intentional breach by Purchaser or the Company of any representation, warranty, covenant or agreement or obligation of such party.

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

EX-2.1 Purchasesoft, Inc.

Chksum: 930327 Cycle 1

Doc 2 Page 28

For all purposes of this Article VIII, the value of Indemnity Shares shall be determined at the time a claim for indemnity is made and shall be the "Fair Market Value" of the Indemnity Shares at that time. For purposes hereof, "Fair Market Value" shall mean the average of the closing price of the common stock of Purchaser as quoted or traded on its primary inter-dealer quotation system or any securities exchange, over the ten (10) trading day period ending three (3) trading days prior to the date on which the subject claim for indemnification is made hereunder. However, if at any time the common stock of Purchaser is not listed on any securities exchange or quoted on an inter-dealer quotation system, "Fair Market Value" shall mean the fair value of the class or series of capital stock of Purchaser constituting Indemnity Shares as determined in good faith and in the sole discretion of a reputable appraiser chosen by the Company's Board of Directors and Purchaser's Board of Directors, together. If these respective Boards of Directors cannot agree on an appraiser within thirty (30) days from the making of the claim, each Board of Directors shall choose a reputable appraiser within ten (10) days thereafter and such appraiser shall have ten (10) additional days to choose the final appraiser, whose appraisal shall be binding on the parties. The cost of such process shall be borne equally by the Company and Purchaser.

8.3 **Restriction on Transferability of the Securities.** The certificates representing the Securities (if and when issued) shall bear the following legend restricting transfer, and such other legends as may be required by any applicable state securities law:

THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF A REGISTRATION STATEMENT IN EFFECT WITH RESPECT TO THE SECURITIES UNDER SUCH ACT OR UNLESS SOLD PURSUANT TO RULE 144 OF SUCH ACT OR ANOTHER AVAILABLE EXEMPTION.

- 8.4 **Agreement not to dispose of Indemnity Shares.** The Company agrees that, for purposes of keeping the Indemnity Shares available for satisfaction of any claims for indemnification pursuant to this Article VIII hereof (if necessary), it will make no transfer, sale or hypothecation of the Indemnity Shares until twelve (12) months following the Closing Date.
- 8.5 **Further Assurances.** In case at any time after the Closing any further action is necessary or desirable to carry out the purposes of this Agreement, each of the parties will take such further action (including the execution and delivery of such further instruments and documents) as the other party reasonably may request, at the sole cost and expense of the requesting party (unless the requesting party is entitled to indemnification therefor under this <u>Article VIII</u>).
- 8.6 **Reports Under Securities Exchange Act of 1934.** With a view to making available to the Company the benefits of Rule 144 promulgated under the Act and any other rule or regulation of the SEC that may at any time permit the Company to sell Purchaser's Common Stock to the public without registration, Purchaser agrees to use its best efforts to:
- (a) make and keep public information available, as those terms are understood and defined in SEC Rule 144, at all times; and
- (b) file with the SEC in a timely manner all reports and other documents required of Purchaser under the Act and the 1934 Act.
- 8.7 **Transfer of Rights to Name.** Subject to approval of the Company's stockholders, the Company shall transfer and assign to Purchaser all rights in the name "Computer Information Enterprises," as the same may be used in whole.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 398743 Cycle 1 EX-2.1 Purchasesoft, Inc.

8.8 Access to Records. After the Closing Date, Purchaser shall retain for a period consistent with Purchaser's recordretention policies and practices those records of the Company relating to the Purchased Assets. Purchaser also shall provide the
Company (including the Company's accountants and auditors) reasonable access thereto, during normal business hours and on at least
three days' prior written notice, to enable them to prepare financial statements or Tax Returns or deal with Tax audits. After the
Closing Date, the Company shall provide Purchaser (including Purchaser's accountants and auditors) reasonable access to records
relating to Excluded Assets, during normal business hours and on at least three days' prior written notice, for any reasonable business
purpose specified by Purchaser in such notice.

8.9 **Transitional Use of Financial System.** The Company may designate one of its employees who will have the right, for a period of 90 days following the Closing, to use (the "Financial System") for the purpose of generating and maintaining the Company's financial reporting obligations and practices. Such employee shall have full use of (and access to) the Financial System for such 90 day period.

ARTICLE IXMISCELLANEOUS

9.1 Termination.

- (a) This Agreement may be terminated and the transactions contemplated hereby abandoned:
 - (i) at any time prior to the Closing, by mutual written consent of Purchaser and the Company;
- (ii) at any time after February 28, 2004 (the "Termination Date"), by the Purchaser or the Company in writing, if the transactions contemplated by this Agreement have not been consummated on or before the Termination Date and such terminating party is not then in material breach of this Agreement; provided that no party may terminate this Agreement pursuant to this clause (ii) if such party's failure to fulfill any of its obligations under this Agreement shall have been a principal reason that the Closing shall not have occurred on or before said date;
- (iii) by the Company on written notice to the Purchaser if (i) there shall have been a material breach of any representations or warranties on the part of Purchaser set forth in this Agreement or if any representations or warranties of Purchaser shall have become untrue in any material respect, provided that the Company has not breached any of its obligations hereunder in any material respect; or (ii) there shall have been a breach by Purchaser of any of its covenants or agreements hereunder in any material respect or materially adversely affecting (or materially delaying) the ability of Purchaser or the Company to consummate the transactions contemplated hereby, and Purchaser has not cured such breach within ten (10) Business Days after notice by the Company thereof, provided that the Company has not breached any of its obligations hereunder in any material respect;
- (iv) by Purchaser on written notice to the Company if (i) there shall have been a material breach of any representations or warranties on the part of the Company set forth in this Agreement or if any representations or warranties of the Company shall have become untrue in any material respect, provided that Purchaser has not breached any of its obligations hereunder in any material respect; or (ii) there shall have been a breach by the Company of any of its covenants or agreements hereunder in any material respect or materially adversely affecting (or materially delaying) the ability of Purchaser to consummate the transactions contemplated hereby, and the Company has not cured such breach within ten (10) Business Days after notice by Purchaser thereof, provided that Purchaser has not breached any of its obligations hereunder in any material respect;

Merrill Corporati	on -04-2032-1 Sat Feb 07 01:03:32 2004	Chksum: 842883 Cy	ycle
EX-2.1	Purchasesoft, Inc.		

(v) by Purchaser, if the Company becomes insolvent or seeks protection under any bankruptcy, receivership, trust deed, creditors arrangement, composition or comparable proceeding, or if any such proceeding is instituted against the Company;

(vi) by Purchaser, if the Company becomes insolvent or seeks protection under any bankruptcy, receivership, trust deed, creditors arrangement, composition or comparable proceeding, or if any such proceeding is instituted against the Company; and/or

(vii) by the Company if the percentage ownership of Purchaser represented by the Closing Shares materially changes between the date hereof and Closing.

- (b) In the event of the termination of this Agreement as provided in this Section 9.1, except as otherwise provided in this Agreement or the instruments and agreements executed in connection herewith, no party shall have any other liability hereunder of any nature whatsoever to any other party, including any liability for Damages; <u>provided</u>, <u>however</u>, that if, at the time of such termination, any party is in default under its obligations hereunder, the party in default shall be liable to the other parties for such default; and <u>provided</u>, <u>further</u>, that the provisions of Section 5.11 and Article IX shall continue in full force and effect.
- (c) In the event that a condition precedent to its obligations is not satisfied, nothing contained herein shall be deemed to require any party to terminate this Agreement, rather than to waive such condition precedent and proceed with the Closing.
- 9.2 **Notices.** All notices, requests and other communications hereunder must be in writing and will be deemed to have been duly given only if delivered personally against written receipt or by facsimile transmission with answer back confirmation or mailed (postage prepaid by certified or registered mail, return receipt requested) or by overnight courier to the parties at the following addresses or facsimile numbers:

If to the Company, to: Computer Information Enterprises, Inc.

Attention: Craig Chandler

EDGAR

Doc 2 Page 30

Facsimile No.:

with copies to: Facsimile No.:

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

If to Purchaser, to: PurchaseSoft, Inc.

7514 Girard Ave Ste 1440

La Jolla, CA 92037

Facsimile No.: 619-839-3793 Attention: Steven Flagg

with copies to:

Facsimile No.: Attention:

All such notices, requests and other communications will (i) if delivered personally to the address as provided in this Section 9.2, be deemed given upon delivery, (ii) if delivered by facsimile transmission to the facsimile number as provided in this Section 9.2, be deemed given upon receipt, and (iii) if delivered by mail in the manner described above to the address as provided in this Section 9.2, be deemed given upon receipt (in each case regardless of whether such notice, request or other communication is received by any other Person to whom a copy of such notice, request or other communication is to be delivered pursuant to this Section 9.2). Any party from time to time may change its address, facsimile number or

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 100514 Cycle 1 EX-2.1 Purchasesoft, Inc.

Doc 2 Page 31

other information for the purpose of notices to that party by giving notice specifying such change to the other parties hereto.

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

9.3 **Entire Agreement.** This Agreement (and all Exhibits and Schedules attached hereto, all other documents delivered in connection herewith and the Confidentiality Agreement) supersedes all prior discussions and agreements among the parties with respect to the subject matter hereof and contains the sole and entire agreement among the parties hereto with respect hereto.

- 9.4 **Waiver.** Any term or condition of this Agreement may be waived at any time by the party that is entitled to the benefit thereof, but no such waiver shall be effective unless set forth in a written instrument duly executed by or on behalf of the party waiving such term or condition. No waiver by any party hereto of any term or condition of this Agreement, in any one or more instances, shall be deemed to be or construed as a waiver of the same or any other term or condition of this Agreement on any future occasion. All remedies, either under this Agreement or by law or otherwise afforded, will be cumulative and not alternative.
- 9.5 **Amendment.** This Agreement may be amended, supplemented or modified only by a written instrument duly executed by or on behalf of each party hereto.
- 9.6 **No Third Party Beneficiary.** The terms and provisions of this Agreement are intended solely for the benefit of each party hereto and their respective successors or permitted assigns, and it is not the intention of the parties to confer third-party beneficiary rights upon any other Person other than any Person entitled to indemnity under Section 8.2 above.
- 9.7 **No Assignment; Binding Effect.** Neither this Agreement nor any right, interest or obligation hereunder may be assigned by any party hereto without the prior written consent of the other parties hereto and any attempt to do so without such consent will be void, except that any party's rights to indemnification under Section 8.2 may be freely assigned. This Agreement is binding upon, inures to the benefit of and is enforceable by the parties hereto and their respective successors and assigns.
- 9.8 **Headings.** The headings used in this Agreement have been inserted for convenience of reference only and do not define or limit the provisions hereof.
- 9.9 **Severability.** If any provision of this Agreement is held to be illegal, invalid or unenforceable under any present or future law, and if the rights or obligations of any party hereto under this Agreement will not be materially and adversely affected thereby, (i) such provision will be fully severable, (ii) this Agreement will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (iii) the remaining provisions of this Agreement will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom and (iv) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Agreement a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible and mutually acceptable to the parties herein.
- 9.10 **Governing Law.** This Agreement shall be governed by and construed in accordance with the laws of the State of California applicable to contracts executed and performed in such State, without giving effect to conflicts of laws principles.
- 9.11 **Consent to Jurisdiction and Forum Selection.** The parties hereto agree that all actions or proceedings arising in connection with this Agreement shall be initiated and tried exclusively in the State and Federal courts located in the County of San Diego, State of California. The aforementioned choice of venue is intended by the parties to be mandatory and not permissive in nature, thereby precluding the possibility of litigation between the parties with respect to or arising out of this Agreement in any jurisdiction other than that specified in this Section 9.11. Each party hereby waives any right it may have to assert the doctrine of forum non conveniens or similar doctrine or to object to venue with

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 584674 Cycle 1 EX-2.1 Purchasesoft, Inc.

EDGAR

Doc 2 Page 32

C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf

respect to any proceeding brought in accordance with this paragraph, and stipulates that the State and Federal courts located in the County of San Diego, State of California shall have in personam jurisdiction and venue over each of them for the purposes of litigating any dispute, controversy or proceeding arising out of or related to this Agreement. Each party hereby authorizes and accepts service of process sufficient for personal jurisdiction in any action against it as contemplated by this Section 9.11 by registered or certified mail, return receipt requested, postage prepaid, to its address for the giving of notices as set forth in this Agreement, or in the manner set forth in Section 9.2 of this Agreement for the giving of notice. Any final judgment rendered against a party in any action or proceeding shall be conclusive as to the subject of such final judgment and may be enforced in other jurisdictions in any manner provided by law.

- 9.12 **Expense.** Each of the parties hereto shall pay the fees, expenses and costs incurred by such party incidental to the preparation of this Agreement and to the consummation of the transactions contemplated hereby.
- 9.13 **Construction.** No provision of this Agreement shall be construed in favor of or against any party on the ground that such party or its counsel drafted the provision. Any remedies provided for herein are not exclusive of any other lawful remedies which may be available to either party. This Agreement shall at all times be construed so as to carry out the purposes stated herein.
- 9.14 **Counterparts.** This Agreement may be executed in any number of counterparts and by facsimile, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

Merrill Corporation -04-2032-1 Sat Feb 07 01:03:32 2004 Chksum: 754281 Cycle 1 EX-2.1 Purchasesoft, Inc.

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the parties hereto, or their duly authorized officer, as of the date first above written.

 $\underline{C:\jms\chanson\04-2032-1\task71588\2032-1-ke.pdf}$

"Purchaser"

PurchaseSoft, Inc.,

A Delaware Corporation

By /s/ Steven A. Flagg

Name: Steven A. Flagg

Title: President and Chief Executive Officer

EDGAR

Doc 2 Page 33

"Company"

Computer Information Enterprises, Inc., a California corporation

By /s/ Craig Chandler

Name: Craig Chandler

Title: President and Chief Executive Officer

28

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004			Chksum: 0 Cycle 1
EX-2.2	Purchasesoft, Inc.		
C:\jms\ugopala\0	4-2032-1\task71564\2032-1-kg.pdf	EDGAR	Doc 3 Header

DOC 3 : Header

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004		Chksum: 761259 Cycle 1
EX-2.2 Purchasesoft, Inc.		
C:\ims\ugonala\04-2032-1\task71564\2032-1-kg pdf	FDGAR	Doc 3 Page 1

Exhibit 2.2

Netshare Solutions GmbH Purchase Agreement

Notarial Deed No. 09/2004

Executed

at Hannover This day of January 30, 2004

appeared before me, the undersigned notary
Harald Wilhelm
In Hannover

the following persons, to No. 1.) identified by official passport:

1. Mr. Werner Kuno **Löchle**, born on December 10, 1950, with business address at Stuttgarterstraße 123, D-71229 Leonberg,

who for the purpose hereof is not acting in his own name, but in the name and on behalf of

PurchaseSoft Inc.,

with its principal place of business in 7514 Girard Avenue, Suite 1440,

La Jolla, CA 92037, U.S.A.

1

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 547253 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 2

on the basis of the attached power of attorney dated January <XX>, 2004.

The notary did not verify the power of the party designated in the power of attorney to act on behalf of the entity set forth therein. Insofar, he is indemnified from any liability.

Known personally:

2. Mr. Gerd **Redöhl**, born on July 12, 1953, residing in Am Walde 5, D-38176 Wendeburg,

who for the purpose hereof is not acting in his own name, but in his capacity as managing director with sole signature acting in the name and on behalf of

TCC Vertriebsgesellschaft mbH

with its principal place of business in Hannover/Germany, registered with the Commercial Register of the Local Court of Hannover/Germany under HRB 57430.

The notary verified proper representation by inspection of the Commercial Register of the Local Court of Hannover.

3. Mr. Dipl.-Ing. Lars **Baumann**, born Steckmann, born on March 9, 1971, residing in An der Schwarzen Riede 20, D-30657 Hannover,

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 962089 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 3

- 4. Mr. Dipl.-Ing. Andreas **Jäckle**, born on August 15, 1966, residing in Wilhelm-Raabe-Weg 10, D-30938 Burgwedel,
- 5. Mr. Dipl.-Ing. Michael **Brusberg**, born on February 12, 1969, residing in Roseggerstraße 8, D-30173 Hannover,
- 6. Mrs. Dipl.-Oec. Tanja Casado **Hempel**, born on September 16, 1972, residing in Windthorstraße 8, D-30167 Hannover.

After asking the parties present, the notary ascertained that there is no prior involvement according to § 3 para. 1 no. 7 of the Notarization Act.

The persons appearing, acting as stated above, asked for the notarization of the following:

AGREEMENT for the Sale and Assignment of Shares

§ 1

Objects of the Agreement

(1) Shareholders of SHA Fünfund-

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 984900 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 4

sechzigste Vermögensverwaltungs-und Beteiligungsgesellschaft mbH (in future named "**netshare solutions GmbH**") with its principal place of business in Hannover/Germany (hereinafter referred to as "**Company**"), registered with the Commercial Register of the Local Court of Hannover under HRB No. 61159, are:

- a) TCC Vertriebsgesellschaft mbH with a capital share in the nominal amount of EUR 10,000.00
- b) Mr. Lars Baumann with a capital share in the nominal amount of EUR 3,750.00
- c) Mr. Andreas Jäckle with a capital share in the nominal amount of EUR 3,750.00
- d) Mr. Michael Brusberg with a capital share in the nominal amount of EUR 3,750.00
- e) Mrs. Tanja Casado Hempel with a capital share in the nominal amount of EUR 3,750.00

Total EUR 25,000.00

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 159823 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 5

The persons listed above under lit. a) through e) are hereinafter collectively referred to as "the Sellers".

The total share capital of the Company amounts to EUR 25,000.00 (in words: Euro twenty-five thousand). The capital shares referred to in sentence 1 above are fully paid-in.

(2) Sellers hereby sell their capital shares in the total amount of EUR 25,000.00 in the Company as referred to in para. 1 above to **PurchaseSoft, Inc.** (hereinafter referred to as "**Purchaser**").

Purchaser hereby accepts such sale.

§ 2

Purchase Price

- (1) The purchase price for the capital shares sold in accordance with § 1 para. 2 above (hereinafter referred to as "**Purchase Price**") shall be
 - a) for the capital share owned by TCC Vertriebsgesellschaft mbH EUR 13,000.00 and the transfer of title of 800,000 shares of

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 964899 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 6

Purchaser's common stock,

- b) for the capital share owned by Mr. Lars Baumann the transfer of title of 1,800,000 shares of Purchaser's common stock.
- c) for the capital share owned by Mr. Andreas Jäckle the transfer of title of 1,800,000 shares of Purchaser's common stock.
- d) for the capital share owned by Mr. Michael Brusberg the transfer of title of 1,800,000 shares of Purchaser's common stock
- e) for the capital share owned by Mrs. Tanja Casado Hempel the transfer of title of 1,800,000 shares of Purchaser's common stock,
- (2) The Purchase Price is due within ten banking days after the notarization of this Agreement. Insofar as the Purchase Price consists of Purchaser's common stock, Sellers shall be provided with respective stock certificates within this period.
- (3) The purchase price for the capital

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 539824 Cycle 1 EX-2.2 Purchasesoft, Inc.

shares sold in accordance with § 1 para. 2 lit. b)-e) above shall be increased for each of these sellers by EUR 7,500.00 respectively,

EDGAR

Doc 3 Page 7

if the cumulative revenue of the Company, calculated since February 1, 2004, reaches

a) EUR 300,000.00, b) EUR 800,000.00, c) EUR 1,440,000.00 and d) EUR 2,000,000.00.

The increase of the purchase price, if any, shall become due within ten banking days after the managing directors of the Company informed Purchaser in writing about the respective cumulative revenue.

§ 3

Effective Date; Right to Receive Profits

The sale of the capital shares to Purchaser becomes effective immediately together with all rights and duties, including the right to receive profits for the current business year and previous years, as far as there are no shareholders' resolutions on the distribution of profits.

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 503574 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 8

§ 4

Assignment

Sellers hereby - subject to full payment of the Purchase Price according to § 2 para. 1 and 2 above and full payment according to § 8 para. 5 below - transfer their capital shares referred to in § 1 para. 1 above in the total nominal value of EUR 25,000.00 to Purchaser with immediate effect. Purchaser hereby accepts such transfer.

§ 5

Declaration of Consent

The shareholders of the Company and Mr. Lars Baumann and Mr. Andreas Jäckle in their capacity as managing directors of the Company approved of the transfer of the capital shares. The respective Shareholders' Resolution and Declaration of Consent are attached hereto as **Exhibits 1 and 2**.

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 273152 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 9

§ 6

Representations and Warranties

The Sellers hereby jointly and severally guarantee, represent and warrant to Purchaser in the meaning of an independent guarantee pursuant to § 311 para. 1 German Civil Code as of the date hereof that:

- a) The information in § 1 above is true and complete in all respects.
- b) The capital shares sold and transferred are free and clear of any rights of third parties, the Sellers are free to dispose of their respective capital shares and this free disposal is not restricted by any rights of pre-emption, option rights or other rights of acquisition of third parties.
 - There are no dormant partnerships, loans coupled with a share in the profits, profit-orientated extra-pay or any other obligations with regard to the results of the Company.
- c) The Company is duly incorporated and validly existing under the laws of Germany and is qualified to transact business in all locations in which it presently transacts business and has power to carry on its business as now being conducted; the Company's Articles of Association ("Satzung") attached as **Exhibit 3** are presently valid and in force and no changes

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 538278 Cycle 1 EX-2.2 Purchasesoft, Inc.

EDGAR

Doc 3 Page 10

have been resolved.

d) The 2003 Annual Report of the Company has been prepared in accordance with applicable laws and accounting principles generally accepted in Germany. The balance sheet in the Annual Report fairly represents the financial position of the Company as of December 31, 2003. The result stated in the income statement in the Annual Report fairly represents the result of the operations of the Company for the period indicated.

- e) Since January 1, 2004, the Company has conducted its business in the normal and ordinary course and has not, other than in the ordinary course of business,
 - $\begin{tabular}{ll} (i) & & disposed of or created any charges or rights over any asset \end{tabular}$
 - (ii) assumed any actual or contingent liability.
- f) Apart from the reservations made in lit. s) below, the Company has good and marketable title to all its assets, free and clear of any liens, encumbrances, pledges or claims.

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 101720 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 11

- g) The Company does not own any real property.
- h) There are no material agreements of the Company other than those entered into in the ordinary course of business and attached as **Exhibit 4**. The entry into this Agreement or any of the transactions contemplated herein will not result in the breach of any of the terms of any agreement which is material to the Company.
- i) All outstanding accounts receivables (less any reserves made for them) will be valid and enforceable claims against customers, for goods or services delivered or rendered in the ordinary course of business, and will be collectible within ninety (90) days after the date hereof.
- j) The Company does not use or require any intellectual property rights other than such as are presently owned or licensed and validly used without infringing any third party's rights; a list of the Company's intellectual property rights is attached as **Exhibit 5**.

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 466818 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 12

- k) The Company does not have any liabilities, fixed or contingent, other than the following:
 - (i) liabilities arising since January 1, 2004 in the ordinary course of business,
 - (ii) liabilities which are to be discharged after the date hereof under any agreement entered into by the Company in the ordinary course of business,
 - (iii) liabilities which are disclosed in this Agreement or any of its Exhibits.
- 1) The Company is covered by insurance according to the details in **Exhibit 6**. Said insurances offer an adequate coverage against all risks normally insured against by companies carrying on the same or a similar business.
- m) **Exhibit 7** contains a list of all persons employed by the Company, specifying their names, titles, rates of compensation and other material fringe benefits. There are no pension liabilities whatsoever.

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 609825 Cycle 1 EX-2.2 Purchasesoft, Inc.

Doc 3 Page 13

n) There are no authorizations given by the Company in force by which any person may enter into any commitment outside the ordinary course of business on behalf of the Company.

o) There are no pending or threatened legal proceedings, litigation, arbitration, administrative proceedings, governmental investigations or claims involving the Company.

 $C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf$

- p) The Company has filed all tax returns required to have been filed by it and has paid or provided for all taxes which have become due or will become due in respect of the period prior to the date hereof. There are no claims or proceedings pending with respect to the Company for past due taxes, and Sellers do not know of any such threatened claims.
- q) The Company has at all times complied with all applicable environmental laws, regulations and licenses and all environmental licenses are valid and subsisting. There is no civil, criminal or administrative action, claim, investigation or other similar proceeding pending against the

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 516506 Cycle 1 EX-2.2 Purchasesoft, Inc.

EDGAR

Doc 3 Page 14

Company from or relating to the breach of any environmental law, regulation or license.

- r) The accounts, registers, financial and other records of the Company have been fully and accurately completed and they are kept, together with all other documents of the Company, at the premises of the Company.
- s) With the exception of the payment of the purchase price, the Company fulfilled all of its obligations under the Contract for the Purchase of Parts of the Business of netshare Aktiengesellschaft, concluded with the insolvency administrator of this company on January 19, 2004; this contract is attached as **Exhibit 8**. After payment of the purchase price, the Company will be unrestricted owner of all assets and rights purchased according to the terms of said contract.
- t) There are no agreements, contracts or transactions of any kind between the Sellers and the Company.
- u) The information contained in the Exhibits and all information given by or on behalf of Sellers to Purchaser or its advisors or agents is true and accurate in all material respects and

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 429393 Cycle 1 EX-2.2 Purchasesoft, Inc.

EDGAR

Doc 3 Page 15

not misleading. All facts concerning the shares and/or business and affairs of the Company which the Purchaser has a reasonable interest to know have been disclosed to him.

§ 7

Indemnification

- (1) The Sellers shall jointly and severally defend, indemnify and hold harmless the Purchaser or, upon Purchaser's choice to be made in writing, the Company from and against any and all loss, damage, cost and expense related to, caused by or arising from any misrepresentation or breach of warranty.
- (2) Purchaser shall be entitled to choose in writing whether Sellers have to compensate Purchaser or the Company by payments or by putting Purchaser or the Company into the same position as if the misrepresentation or breach of warranty had not occurred or existed.

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 902790 Cycle 1 EX-2.2 Purchasesoft, Inc. $C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf$ **EDGAR**

§ 8

Doc 3 Page 16

Additional Covenants

- Sellers are aware that Purchaser's common stock acquired as part of the Purchase Price is subject to Rule 144 of the US-SEC (1) (Securities and Exchange Commission); Sellers hereby undertake to make any and all additional declarations that might be necessary to effectively acquire legal title to these shares of Purchaser.
- Mr. Andreas Jaeckle shall resign from his post as managing director of the Company immediately upon notarization of this (2) Agreement and shall be employed with the Company according to the employment contract attached as Exhibit 9, thereby canceling his current managing director contract.
- (3) Mr. Lars Baumann shall continue to act as managing director of the Company according to the terms of a managing director contract to be negotiated between the Parties within 30 days after the notarization of this Agreement.
- Mr. Michael Brusberg and Mr. Lars Baumann provided to Sparkasse Hannover security in the amount of EUR 100,000.00 for (4) a loan granted to

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 265795
 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 17

- netshare AG. If Sparkasse Hannover recourses to this security, Purchaser shall assume 50 % of this security, but in no case more than EUR 50,000.00, on terms still to be agreed between the Parties.
- (5) Purchaser shall provide the Company upon Purchaser's choice either with a working capital loan or a payment into the capital reserve in the amount of EUR 187,000.00; payment is due immediately after notarization of this Agreement.
- (6) Purchaser shall provide Sellers with a confirmation of its company secretary within 30 days after notarization of this Agreement, stating that the Board of Directors of Purchaser entirely approved of the conclusion of this Agreement.

§ 9

Limitation Period

Any claims of Purchaser according to § 7 of this Agreement shall be fully effective and enforceable until December 31, 2006. The limitation period will be interrupted by a

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004 Chksum: 916163 Cycle 1 EX-2.2 Purchasesoft, Inc.

EDGAR

Doc 3 Page 18

written notification to Seller; however, the limitation period will lapse if Purchaser does not file an action with an appropriate court until June 30, 2007 at the latest.

§ 10

Confidentiality

- (1) The Parties agree not to disclose to any third party the subject matter or terms of this Agreement. This obligation can only be terminated by mutual consent. Any publication of the subject matter or terms of this Agreement requires the written consent of all Parties.
- (2) This obligation does not apply to any duty to disclose by law or administrative rule; in addition, the Parties are free to provide their respective advisors which are bound by professional rules of secrecy with all necessary information. Moreover, Purchaser is exempted from the obligation of confidentiality according to para. 1 above with respect to its representatives, shareholders and their representatives.

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 643075
 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 19

§ 11

Costs and Taxes

- (1) Taxes on transaction profits, if any, shall be borne by the respective Sellers.
- (2) The Parties shall equally bear the costs connected with the notarization of this Agreement.
- (3) Each Party bears its own costs, especially the costs of its advisors.

§ 12

Final Provisions

- (1) This Agreement shall be subject to German law. Any dispute relating to this Agreement shall be exclusively submitted to the Courts of Hannover.
- (2) For any claims under this Agreement the German version shall prevail. The English version is only a translation of

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 970025 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 20

the German version.

- (3) This Agreement contains the entire agreement and understanding among the Parties with respect to the transactions contemplated hereby. Any amendment with respect to any provision of this Agreement shall not be effective unless the same shall be notarized.
- (4) Should any provision of this Agreement be invalid in whole or in part, this shall not affect the validity of the remaining contractual provisions. The same applies in case of a contractual gap. In lieu of the invalid provision or as a provision filling the gap, an appropriate provision shall be deemed agreed, which, to the extent legally possible, comes as close as possible to what the contracting parties would have agreed upon, had they taken this aspect into account when drafting this Agreement.
- (5) The notary is asked to provide copies of this deed as follows:
 - a) the Parties:

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:24:23 2004
 Chksum: 668126 Cycle 1

 EX-2.2
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf
 EDGAR
 Doc 3 Page 21

one certified copy each,

- b) the Company: one certified copy (as notification according to § 16 GmbHG),
- c) the Fiscal Authorities: one certified copy each,
- d) Dr. Matthias Döring, c/o Schrade & Partner, Bismarckallee 15, 79098 Freiburg:
- e) Herr Werner Kuno **Löchle**, Stuttgarterstraße 123, 71229 Leonberg, one uncertified copy.

The above document have been read in the presence of the notary. The parties agreed not to read the exhibits in detail. It has been approved of by the persons appearing, acting as indicated, and personally signed as follows:

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004		Chksum: 24676 Cycle 1
EX-2.2 Purchasesoft, Inc.		
C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf	EDGAR	Doc 3 Page 22
PurchaseSoft, Inc.	Andreas Jäckle	_
		_
TCC Vertriebsgesellschaft mbH	Michael Brusberg	_
	22	

Merrill Corporation -04-2032-1 Sat Feb 07 10:24:23 2004		Chksum: 673504 Cycle 1
EX-2.2 Purchasesoft, Inc.		
C:\jms\ugopala\04-2032-1\task71564\2032-1-kg.pdf	EDGAR	Doc 3 Page 23
Lars Baumann	Tanja Casado	
	23	

Merrill Corporation -04-2032-1 Sat Feb 07 10:25:51 2004 Chksum: 110199 Cycle 1

EX-99.1 Purchasesoft, Inc.

Exhibit 99.1

Press release: Source: Purchasesoft, Inc.

Thursday February 5, 2004

Purchasesoft Acquires Netshare Solutions of Hanover, Germany

LA JOLLA, CA.

PurchaseSoft, Inc. (OTC: Pink Sheets PURC.PK) today announced it has acquired Netshare Solutions, GmbH for cash, stock, and the assumption of certain liabilities. Netshare Solutions is an innovative solutions provider, specializing in eprocurement and eSales solutions in the B2B (Business-to-Business) and B2C (Business-to-Consumer) environment.

Steve Flagg, PurchaseSoft Chairman and CEO stated, "We are excited about this acquisition. Netshare has award winning products, a solid customer base and incredible partnerships. We will continue to look for strategic acquisitions to build the company into a robust provider of Enterprise solutions. Netshare is a classic example of the type of acquisition we will pursue in the future."

Lars Baumann, Managing Director of Netshare Solutions, stated, "Becoming a PurchaseSoft company gives Netshare Solutions GmbH great future opportunities. With the experience of PurchaseSoft we can get access to international markets. Our award winning product FlexTrade will now be offered in the US market. The joint experience will accelerate the growth of the company. With the strong financial background of PurchaseSoft we can provide our European clients more security for their future investments in our software. In one year from now I expect Netshare to be a totally different company with a strong international approach" Werner Loechle, COO PurchaseSoft stated "The procurement market is far from saturated. We are positioned to benefit from the fast consolidation process in the industry, many suppliers will disappear, and clients need the continuity we offer to secure their investment decisions. Our business model now includes license, ASP, and service revenue, we are industry-independent and cover small companies to blue-chip players from local to global and finally we have the support of Microsoft, the world's largest software company due to our technology. The addition of Netshare adds depth and talent to our highly skilled management team".

About PurchaseSoft: Founded in 1977, PurchaseSoft provides a complete product suite to address enterprise procurement requirements from requisitioning to analytics and sourcing. The company draws on its historical expertise in purchasing to provide robust solutions to small and mid-size businesses.

This press release contains forward-looking statements relating to, among other things, the expected benefits of the Company's recent acquisitions. Such statements reflect management's best judgment based on factors currently known but are subject to risks and uncertainties, which could cause actual results to differ materially from those anticipated. Such risks and uncertainties include, but are not limited to, the Company's ability to finalize large client contracts, fluctuations in customer demand for the Company's products and services, intensity of competition from application vendors, delays in product development, the Company's ability to

 Merrill Corporation -04-2032-1
 Sat Feb 07 10:25:51 2004
 Chksum: 947446 Cycle 1

 EX-99.1
 Purchasesoft, Inc.

 C:\jms\ugopala\04-2032-1\task71564\2032-1-mm.pdf
 EDGAR
 Doc 4 Page 2

control expenses, general economic and industry conditions, terrorist activities, the Company's ability to integrate acquired businesses, the effect of the acquisitions on customer demand for the Company's products and services, and those risks described in the Company's filings with the Securities and Exchange Commission, PurchaseSoft cautions investors that it may not update any or all of the foregoing forward-looking statements.